## Western University Scholarship@Western

**Electronic Thesis and Dissertation Repository** 

2-22-2024 1:00 PM

# Characterizing Hypoxia, Neutrophil Persistence and Revascularization in the Murine db/db Model of Type II Diabetic Impaired Skin Healing

Michael R. Grynyshyn, Western University

Supervisor: Hamilton, Douglas W., *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Master of Science degree in Anatomy and Cell Biology © Michael R. Grynyshyn 2024

Follow this and additional works at: https://ir.lib.uwo.ca/etd

🔮 Part of the Cell Biology Commons, and the Laboratory and Basic Science Research Commons

#### **Recommended Citation**

Grynyshyn, Michael R., "Characterizing Hypoxia, Neutrophil Persistence and Revascularization in the Murine db/db Model of Type II Diabetic Impaired Skin Healing" (2024). *Electronic Thesis and Dissertation Repository*. 10020. https://ir.lib.uwo.ca/etd/10020

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

### Abstract

Impaired skin healing represents a significant clinical burden. In the diabetic, inflammatory aberrations, hypoxia and insufficient angiogenesis all result in negative wound healing outcomes - repeated infections, poor perfusion and ultimately amputation. Previous research has reported comparable levels of neutrophils in closed wounds up to 4-12 weeks old. Our study focus was in investigating the dynamics of hypoxia resolution, neutrophil persistence and angiogenic response in the db/db model. Contrary to our hypothesis, we observed significantly higher hypoxic load in the wild types at days 3 and 7. Additionally, we observed significantly elevated neutrophil numbers at day 7 db/db wound bed and an angiogenic deficit at day 3, with wild types exhibiting significantly more CD31<sup>+</sup> cells. Our results validate the db/db model as one of impaired healing, and are consistent with literature in the field suggesting a potential deficit in hypoxic adaptation resulting in an overall delayed wound healing process.

Keywords: hypoxia, inflammation, neovascularization, wound-healing

ii

## **Co-Authorship Statement 1**

The pig as a model system for investigating the recruitment and contribution of myofibroblasts in skin healing.

Douglas Hamilton, John Walker, Dylan Tinney, Michael Grynyshyn, Alexander El-Warrak, Emily Truscott, Lauren E. Flynn

Publication Status: Published January 2022, Wound Repair and Regeneration, Volume 30, Issue 1, pages 45-63

Citation: Hamilton DW, Walker JT, Tinney D, Grynyshyn M, El-Warrak A, Truscott E, Flynn LE. The pig as a model system for investigating the recruitment and contribution of myofibroblasts in skin healing. Wound Repair Regen. 2022 Jan;30(1):45-63. doi: 10.1111/wrr.12981. Epub 2021 Nov 6. PMID: 34708478.

Personal Contribution:

- Wrote the introduction section
- Performed and imaged the a-SMA tissue processing Figure 1 and Figure 2

Estimated Personal Contribution Percentage: 20%

## **Co-Authorship Statement 2**

Microcirculation surrounding end-stage human chronic skin wounds is associated with endoglin/CD146/ALK-1 expression, endothelial cell proliferation and an absence of p16<sup>Ink4a</sup>

Jiarong Wang, Dylan Tinney, Michael Grynyshyn, J. Geoffrey Pickering, Adam Power, Luc Dubois, Douglas Hamilton

Publication Status: Published April 2023, Wound Repair and Regeneration, Volume 31, Issue 3, Pages 321-337

Citation: Wang J, Tinney D, Grynyshyn M, Pickering JG, Power A, Dubois L, Hamilton DW. Microcirculation surrounding end-stage human chronic skin wounds is associated with endoglin/CD146/ALK-1 expression, endothelial cell proliferation and an absence of p16<sup>Ink4a</sup>. Wound Repair Regen. 2023 May-Jun;31(3):321-337. doi: 10.1111/wrr.13081. Epub 2023 Apr 13. PMID: 37017097.

Personal Contribution:

- Imaged the CD31 tissue sections in Figure 1
- Imaged the CD146 tissue sections in Figure 2
- Imaged the Endoglin tissue sections in Figure 3
- Performed and imaged the Endoglin/CD146 processing in Figure 4
- Imaged part A) in Figure 6
- Sent tissue to university hospital and imaged it in Figure 9

Estimated Personal Contribution Percentage: 15%

### **Summary for Lay Audience**

Type II Diabetes is a disease that arises from the body's inability to deliver sugar into cells effectively. The result is increased levels of sugar in the blood, often for a very prolonged period of time. Diabetics are also at increased risk for wounds, especially on their feet due to issues with their blood vessels and nerves, which result in poor sensation. Once a wound is created, it often starts a cycle of trauma, inflammation and repeated infection. Diabetics are at a greatly increased risk for foot amputation, and the prognoses following these operations is often poor.

Hypoxia, inflammation and new blood vessel formation are all well-known factors in wound healing. The purpose of this work is to utilize a diabetic and healthy mouse model, to attempt to outline any differences between them with respect to these parameters. The use of such models allows researchers to model complex pathologies in a relatively quick and inexpensive manner. We seek to uncover differences between healthy and diabetic mice in order to further our understanding of what is different between how a diabetic and a healthy person and potentially extrapolate these findings to the discovery of new therapeutics.

Hypoxia, inflammation and vascular are all attractive and promising areas of research for the development of new therapeutics. In the present study we aimed to outline differences in how quickly hypoxia resolves, and whether we could establish a relationship between its resolution, the amount of inflammation present in the tissue and the amount of new blood vessel ingrowth into the wound.

### Acknowledgments

I would like to acknowledge my supervisor Douglas Hamilton, for giving me a chance in his laboratory. Without his guidance, patience and support I would not be where I am today. Doug taught me how to write in a scientific manner, and instilled in me the importance of attention to detail. His guidance in terms of experimental design was also invaluable in my project.

I would like to acknowledge and thank Sarah Brooks, Georgia Nikoloudaki, Alex Peidl and JT Walker for teaching me practically every scientific technique I now know, from cryosectioning and IHC, to mouse surgeries. Without Sarah and Georgia in particular, I also would not be where I am today. They were like big sisters to me in the lab. I would also like to acknowledge and thank Aishik Chakraborty for spending time with me to train me on how to make figures with ease in Illustrator. I acknowledge everyone I have worked with in Hamilton Lab, I have been able to learn something from each of you. Lastly, great thanks go out to all ACVS staff who often helped with last minute surgery supplies and booking OR rooms.

I would also like to acknowledge and thank my committee for providing continuing guidance to me throughout my degree. Western University has been a great institution for the pursuit of my degree.

vi

# **Table of Contents**

Abstractii
Co-Authorship Statement 1iii
Co-Authorship Statement 2iv
Summary for Lay Audiencev
Acknowledgmentsvi
Table of Contents
List of Figuresix
List of Appendicesx
List of Abbreviationsxi
Chapter 11
Introduction1
1.1 Mechanism of Initial Ulceration3
1.2 Inflammation: What is Normal in Skin Healing?3
1.3 Diabetic Foot Ulcer Inflammation: What is Wrong?5
1.4 The Rogue Agent – Neutrophil6
1.5 Neutrophil Persistence in Wound Healing – A Different Perspective8
1.6 Model Validation10
1.7 The Macrophage11
1.8 Angiogenesis: What is Considered Normal?12
1.9 Diabetic Angiogenesis: What is Wrong?14
1.10 Rationale
Chapter 2 17
Characterizing Hypoxia, Neutrophil Persistence and Revascularization in the Murine db/db Model of Type II Diabetic Impaired Skin Healing17
2.1 Introduction & Significance19
2.2 Materials and Methods 20   2.2.1 Mice 20   2.2.2 Excisional Wounding Experiments 20
2.2.3 Human & Porcine Tissue Samples 21   2.2.4 Histology 22

2.2.5 Cell Counting Quantifications	23
2.2.6 Hypoxyprobe <sup>TM</sup> Assay	24
2.2.7 Statistical Methods	24
2.3 Results	25
2.3.1 Diabetic mice exhibit a delay in the rate of wound contraction, re-epithelialization and	collagen
deposition.	25
2.3.2 – Wild-type wounds are qualitatively more hypoxic at day 7 post-wounding	
2.3.3 There are no significant differences in the amount and persistence of epithelial hypoxia	a
between phenotypes at Days 3 & 7.	37
2.3.4 Wild-type wounds present with more hypoxic cells in their wound beds and edge at Da more hypoxic cells in wound beds at Day 7.	iy 3 and 44
2.3.5 There are no significant differences in neutrophil load at Day 3. Diabetic wounds prese	ent with
a higher neutrophil load in their wound beds at Day 7.	51
2.3.6 Wild-type mice exhibited significantly more CD31 <sup>+</sup> cells at Day 3 in their wound beds edges. There were no significant differences in amount of CD31 <sup>+</sup> blood yessels at Day 7 betw	and right veen
nhenotypes.	
2.3.7 Neutrophil elastase staining in both human and porcine tissue	65
2 5 Deferences	00
2.5 Keterences	
Chapter 3	88
General Discussion	
3.1 Summary	88
3.2 Contributions to the Literature	90
3.2.1 Oxygen and the Healing Wound	90
3.2.2 Tools for Quantifying Cellular Hypoxia: Hypoxyprobe <sup>TM</sup>	91
3.2.3 Hypoxia and Tissue Response During Skin Healing	93
3.2.4 Hypoxia and Inflammation	
3.2.5 Hypoxia and Angiogenesis	
3.2.6 Murine Model and it's Validation	
3.2.7 Future Directions	109
3.2.8 References:	112
4.0 Appendix	120
5.0 Curriculum Vitae	143

## List of Figures

Figure 1 – Diabetic mice exhibit a significant delay in closure kinetics 34
Figure 2 – Diabetic mice exhibit delayed re-epithelialization
Figure 3 – Picrosirius Red stain showing no differences in collagen deposition
at day 3
Figure 4 – Diabetic mice exhibit decreased collagen deposition 41
Figure 5 – General Hypoxyprobe <sup>TM</sup> staining pattern
Figure 6 – Epithelial Hypoxia Distance – Day 3 & 7 46
Figure 7 – Hypoxic Cell Load – Day 3 & 7 53
Figure 8 – Neutrophil Load – Day 3 & 7 60
Figure 9 – CD31 count – Day 3 & 7 67
Figure 10 – Human Neutrophil Elastase 75
Figure 11 – Porcine Neutrophil Elastase Day 7 & 28

# List of Appendices

Appendix 4.1 – Hypoxyprobe <sup>TM</sup> Stain Validation 136
Appendix 4.2 – Neutrophil Elastase Stain Validation 138
Appendix 4.3 – CD31 Stain Validation 140
Appendix 4.4 – Total Cell Counts, Wild Type 142
Appendix 4.5 – Total Cell Counts, Diabetic 144
Appendix 4.6 – "Volcano like" cell infiltration pattern 146
Appendix 4.7 – Positive Control for Hypoxyprobe <sup>TM</sup> 148
Appendix 4.8 – Temporal Hypoxia Resolution Day 12 in Wild Type150
Appendix 4.9 – Hypoxyprobe <sup>TM</sup> staining in tumor 152
Appendix 4.10 – Unwounded Murine Hypoxyprobe <sup>TM</sup> Staining 154
Appendix 4.11 – Animal Protocol Approval Letter

## **List of Abbreviations**

- AGE advanced glycation end product
- ANG -1 angiopoetin 1
- ANG 2 angiopoetin 2
- ATP adenosine triphosphate
- CD31 cluster of differentiation 31
- DAB diaminobenzene
- DFU diabetic foot ulcer
- ECM extracellular matrix
- $FGF-fibroblast\ growth\ factor$
- HIF 1 hypoxia inducible factor 1
- HClO-hypochlorous acid
- IL interleukin
- miR210 microRNA 210
- MMP matrix metalloproteinase
- NF-kB nuclear factor kappa beta
- PEDF pigment epithelium derived factor
- ROS reactive oxygen species
- SEM scanning electron microscopy
- SPRY2 sprouty 2
- TGF-B transforming growth factor beta
- TIMP tissue inhibitor of matrix metalloproteinases

TNF-a – transforming growth factor alpha

VEGF-vascular endothelial growth factor

## **Chapter 1**

### Introduction

The importance of proper wound healing has been recognized as conducive to good health since the dawn of man. Papyrus scrolls dating back to 3200-3000 BC, outlining techniques of compression and hemostasis, point to the recognition of importance of facilitating this process by early medical practitioners. Up until about 100 years ago the biggest health problem facing humanity was infectious disease, owing to issues of sanitation and overcrowding, while further compounded by a lack of available treatments. With the advent of vaccines and antibiotics, we were able to successfully overcome these issues, shifting the current medical focus to addressing other sorts of debilitating, chronic conditions such as cancer, AIDS, neurodegenerative disorders and diabetes. The focus of this manuscript will be Diabetes Mellitus, with a specific emphasis on its commonly seen complication – diabetic foot ulcers (DFUs). Before beginning the comprehensive discussion of the DFU pathology, it is necessary to preface with a brief discussion of the types of chronic wounds and what their medical definition is.

Chronic wounds are defined as wounds exhibiting a disrupted repair process where a sustained anatomical and functional result is not reached within three months <sup>[1]</sup>. The health issues that often result in chronic wounds are rapidly increasing in most parts of the world, thus clinicians can expect to see an increase in the incidence of non-healing pressure, venous and diabetic ulcers. The majority of chronic wounds fall into three main categories: venous ulcers, pressure ulcers and diabetic ulcers <sup>[2]</sup>. Venous ulcers represent more than half of all lower limb chronic wounds and will affect 1-2% of the adult population, with a higher prevalence in women and

the elderly <sup>[3]</sup>. Pressure ulcers tend to afflict those with compromised mobility or sensory perception. Patients who are paralyzed or unconscious for long periods of time cannot feel stimuli which would otherwise tell them it might be time to reposition. Such prolonged unrelieved pressure can lead to ischemia when tissue compression exceeds capillary pressure <sup>[4]</sup>. Regardless of their etiology, chronic wounds are all characterized by enhanced local hypoxia and inflammation. Sadly, the significance of chronic wounds is often overshadowed by their causes, the costs are poorly documented and appropriate care and education is lacking <sup>[5]</sup>. Chronic wounds persist as a silent epidemic, adversely impacting the quality of life of over 40 million individuals worldwide <sup>[6]</sup>.

Type 2 diabetes is a metabolic disease, characterized by prolonged elevated blood sugar levels, resulting from the body's inability to effectively shuttle it into cells. There are currently over 382 million people afflicted with this condition worldwide <sup>[7]</sup>. Unfortunately, due to factors such as an increasingly aging population and high fat diets common in Western culture, these numbers are projected to rise. Diabetic foot ulcers are a commonly seen and serious complication of diabetes <sup>[8]</sup>, with approximately 15% of patients developing one. Of those that develop a DFU, 14-24% will subsequently undergo a lower limb amputation, with a five-year post-op mortality rate from this procedure approaching 50-59% <sup>[9–11]</sup>. It is clear that therapeutic strategies for promoting the closure of DFU's warrant investigation, due to them being a common and costly complication, and the burden they impart on the patient's life and the healthcare system.

#### 1.1 Mechanism of Initial Ulceration

In diabetics, a pathogenic triad of neuropathy, ischemia, and trauma is traditionally described <sup>[12]</sup>. A common denominator across DFU's is poor vascular flow, a condition that has been repeatedly implicated in poor wound healing <sup>[13]</sup>. Peripheral neuropathy results in reduced sensation of the foot, which is often occurring in tandem with reduced perfusion. When this poorly perfused, insensate foot is repeatedly stressed in ways not natural to it, the risk of skin damage and ulceration increases <sup>[14]</sup>. Once the initial break in the skin occurs, an insidious cycle of repetitive trauma and inflammation begins. This cycle is called ischemia-reperfusion, and it has been proposed to be the precipitating event for several other types of chronic wounds of differing etiologies <sup>[15]</sup>. Diabetic patients with vascular deficiencies are subjected to cyclic periods of hypoxia in their lower limbs, during activities such as standing. These are followed by periods of repetfusion, when their limbs are elevated and pressure is relieved.

Hypoxia has a strong potential to induce and maintain a pro-inflammatory state, so during periods of reperfusion the patient's wound sees an influx of leukocytes, which produce pro-inflammatory cytokines and reactive oxygen species (ROS), adding to the already accumulated ROS from partial tissue reoxygenation <sup>[16]</sup>. Nitric oxide – a vasodilator, is downregulated, further exacerbating the inflammatory response <sup>[17]</sup>. It is evident that as long as the patient keeps attempting to use their feet, these cycles are repeated, along with the deleterious effects of prolonged, repetitive bouts of hypoxia induced inflammation which will be outlined next.

### 1.2 Inflammation: What is Normal in Skin Healing?

To begin a discussion of the perpetually inflamed DFU pathology, it is necessary to briefly outline the process of normal (acute) inflammation. Upon cutaneous injury, circulating *MSc. Thesis – April 2024* 3

platelets are exposed to the underlying extracellular matrix (ECM) and are induced to aggregate into a fibrin clot. They also release platelet derived growth factor (PDGF), which plays an important role in initiating the influx of neutrophils, monocytes, smooth muscle cells and fibroblasts to the wound bed. Both platelets and infiltrating leukocytes release interleukins (IL) IL-1a, IL-1B, IL-6, IL-8, Tumor Necrosis Factor Alpha, Platelet Derived Growth Factor and Transforming Growth Factor Beta (TGF-B), the latter of which stimulates cytokine release from macrophages and enhances fibroblast and smooth muscle cell migration <sup>[18]</sup>. The influx is further facilitated by degranulated mast cells, which release histamine, creating pores in blood vessels, thus aiding extravasation of leukocytes to the wound site and facilitating protein leakage.

The initial leukocyte response is mainly characterized by neutrophils for the first 2-5 days, with macrophages replacing them at approximately day 3. Neutrophils have three main functions: 1) Generate ROS via myeloperoxidase pathway to kill pathogens, 2) Debride the wound by breaking down nonviable tissue with the help of various proteolytic enzymes, 3) Phagocytose dead bacteria and matrix debris <sup>[19]</sup>. Once these processes are carried out, neutrophils undergo apoptosis and are cleared out by macrophages. This "clearing out" process is termed efferocytosis, and it prevents secondary necrosis of neutrophilic cells, and is thought to be essential for complete tissue repair <sup>[20]</sup>.

As monocytes migrate into the wound, they mature into macrophages and become one of the most important regulatory cells in the inflammatory cascade <sup>[21]</sup>. They remove any nonfunctional host cells, bacterial-filled neutrophils (efferocytosis), damaged matrix, foreign debris, and remaining bacteria<sup>[22]</sup>. Additionally, they release growth factors and cytokines such as Transforming Growth Factors Alpha and Beta, basic Fibroblast Growth Factor and Vascular Endothelial Growth Factor (VEGF) in order to amplify and eventually resolve inflammation. MSc. Thesis – April 2024 4

Some of these secretions serve to recruit endothelial cells and fibroblasts to progress the wound to the proliferative phase.

Under normal circumstances, inflammation resolves within 1-2 weeks of injury, when the leukocyte numbers fall back to their pre-inflammation numbers and phenotypes <sup>[23]</sup>. Unfortunately, in the presence of a deleterious stimuli, such as persistent hypoxia, accumulation of advanced glycation end products, or repeated trauma, inflammation can become prolonged and/or exacerbated. Additionally, diabetes mellitus leads to hyperglycemia-related metabolic changes that directly impair wound healing <sup>[24]</sup>. A central component of these changes is accumulation of advanced glycation end-products (AGEs) which can induce oxidative stress, impair skin and inflammatory cell function, increase ECM stiffness, and perhaps most importantly – induce a chronic low-grade, self-perpetuating inflammatory state along with circulatory dysfunctions leading to poor tissue oxygenation <sup>[25–28]</sup>.

#### 1.3 Diabetic Foot Ulcer Inflammation: What is Wrong?

DFU's are known to be locked in a perpetual cycle of inflammation, unable to progress to proliferative and remodeling phases of wound healing <sup>[29]</sup>. It is also known that several types of chronic wounds occur in the background of local tissue hypoxia due to various vasculopathies <sup>[30]</sup>. Although the molecular mechanisms vary, generally, hypoxia leads to cell membrane disruption and promotion of deleterious inflammatory cascades <sup>[31]</sup>. The general relationship between hypoxia and inflammation can be described as mutually reinforcing. It is known that in a healthy person, acute cutaneous injury results in temporary local tissue hypoxia, which is eventually resolved by the orderly phases of wound healing and angiogenesis. Conversely, in a diabetic individual, the local hypoxia in the DFU is never truly resolved, as a result of a number of factors which will be outlined next.

It is established that the accumulation of AGEs results in a chronic, low grade inflammatory state for the diabetic patient. It is also known that hypoxic zones tend to arise in inflamed tissues, as inflammation is intricately linked to oxygen metabolism <sup>[32]</sup>. Heat, swelling and redness - the three classical manifestations of inflammation, result from enhanced blood flow and vascular permeability and are thus directly associated with reduced oxygen distribution in inflamed areas. It must be noted that enhanced blood flow, while being suggestive of enhanced oxygen delivery, is anything but so. This phenomenon is attributed to reduced oxygen diffusion due to higher interstitial pressure (swelling) and enhanced oxygen consumption by cells as they attempt to cope in the harsh, inflamed environment. While inflammation creating a metabolically demanding environment for cells is deleterious in itself, an additional caveat is that hypoxic conditions in and of themselves are able to induce inflammatory reactions <sup>[33]</sup>. Studies have shown that mice exposed to 5% O<sub>2</sub> for 60 minutes exhibited significantly enhanced protein expression of interleukin (IL)-6, tumor necrosis factor alpha (TNF-a), and interleukin (IL)-1 in both serum and isolated macrophages. Similar observations were made in healthy human volunteers who showed increased serum levels of proinflammatory factors after three overnight stays at high altitude <sup>[34]</sup>. This information lends credibility to the idea that hypoxia and inflammation are closely related and are capable of mutually reinforcing one another<sup>[35]</sup>.

### 1.4 The Rogue Agent – Neutrophil

The cell that seems to draw the suspicion of most investigators looking at pathologically prolonged inflammation in the context of hypoxia is – the Neutrophil. Upon detection of bacteria or mediators of inflammation, neutrophils begin phagocytosing microbes, which is followed by an assembly of an electron transport chain (NADPH oxidase) which delivers

electrons across the membrane to molecular oxygen for the generation of hypochlorous acid (HClO) and ROS leading to the lysis of microbes <sup>[36]</sup>. This process is called "respiratory burst", and it requires a significantly elevated consumption of molecular oxygen <sup>[37]</sup>. The respiratory burst accounts for an essential antimicrobial function, allowing neutrophils to carry out their decontamination tasks effectively. In summary, the presence of activated neutrophils at sites of inflammation results in oxygen depletion, a phenomenon aptly termed "inflammatory hypoxia" <sup>[38]</sup>. While neutrophils are known to be abundant in the early stages of wound healing, and are known to be essential for effective decontamination, a large, body of emerging evidence implicates their overabundance in impaired wound repair; as their persistence has the potential to initiate processes which destroy healthy tissue <sup>[39]</sup>.

Perhaps the most detrimental potential consequence of prolonged hypoxia/inflammation, is the increased survival time of neutrophils <sup>[40]</sup>. Hypoxia has been shown to inhibit neutrophilic apoptosis via nuclear factor kappa beta (NF-kB) signaling, thus marking NF-kB as a regulator of hypoxic response in neutrophils. Additionally, the neutrophil activating/survival factor MIP-1B was shown to be induced under hypoxic conditions operating as an alternative mediator to neutrophil survival <sup>[41]</sup>. When this increased survival rate is considered together with the neutrophils ability to shape the tissue microenvironment via depletion of local molecular oxygen and degradation of ECM, the negative implications for sustained hypoxia become clear.

There have been numerous studies implicating increased neutrophil influx/persistence in aggravated disease outcomes. In order to carry out their debridement functions, neutrophils carry cationic peptides, eicosanoids and proteinases. Proteases such as elastase and cathepsin G, are capable of degrading most components of the ECM as well as proteins such as clotting factors, cytokines and immunoglobulins. It is well established that the ECM is an essential *MSc. Thesis – April 2024* 7

component in the wound microenvironment, serving as a scaffold for infiltrating cells, so any adverse neutrophil modifications to it may have profound consequences for wound repair. Neutrophil overabundance is also associated with an over-production of ROS, causing further damage to the ECM, cell membranes and leading to premature cell senescence <sup>[42]</sup>. Additionally, neutrophils release serine proteases and matrix metalloproteinases (MMPs) such as neutrophil collagenase; elastase degrades important growth factors such as PDGF and TGF-B while collagenase degrades and inactivates components of the ECM. In addition to ECM damage, neutrophil secreted pro-inflammatory cytokines IL-1B and TNF-a not only increase production of matrix metalloproteinases, but also reduce their tissue inhibitors (TIMPs). This imbalance further augments the degradation of ECM, impairs cell migration, and reduces fibroblast proliferation and collagen synthesis.

#### 1.5 Neutrophil Persistence in Wound Healing – A Different Perspective

A considerable deal of supporting rationale for this project came from the work of Rosalind Butterworth, MD, in 1992. Although the work may seem dated, this must be considered in context of one of the biggest obstacles to human wound research – a substantial lack of human models on which research can be conducted. Although human studies have been carried out since 1992, there is still a recognized lack of detailed information regarding the histological differences between chronic and healthy human wounds. Naturally, this brings up the issue of what it is investigators are trying to "model" in their animal wound healing studies. If sufficient understanding of the differences in cell populations and overall histological presentation has not been attained in human wounds, it can conceivably create difficulty when trying to extrapolate findings and data from animal models. Thus, the starting point for our project was

model validation, as the histological presentation of human wounds differs markedly from the histological findings in similar wounds in laboratory animals.

The primary aim of the Butterworth thesis was to investigate the structure and cellularity of human granulation tissue, via taking punch biopsies from healthy and unhealthy wounds. A secondary aim was evaluation of commonly used animal models of wound healing, and the extent to which data from them can be applied to the understanding of the human wound. There was a total of 24 healthy wounds, and 14 unhealthy wounds assessed, the latter of which were separated into three categories – overgranulating wounds, infected wounds and non-healing wounds. The most striking findings to come out of her research, which were used to form the partial rationale for this project were:

- When counts of all cell types from healthy wounds were considered together, there was a striking persistence of neutrophils and inflammatory cells throughout healing in closed wounds up to 4-12 weeks old.
  - It is normal in healthy human wounds to find an infiltrate of inflammatory cells including neutrophils, macrophages and lymphocytes whose distribution remains similar throughout healing.
- Overall distribution of inflammatory cells in infected or overgranulating wounds was similar to that of healthy wounds.
- The almost complete lack of inflammatory cells observed in non-healing wounds.
- Of the seven cell types examined only capillary endothelial cell number was found to be related to good healing progress.

These findings stand in contrast with the general consensus among many investigators that neutrophil persistence in the context of wound healing is a detrimental outcome. There has been a considerable amount of work done in attempting to understand the function of this cell, and the potential negative implications associated with its persistence. The main body of the data suggests that beyond the initial acute inflammatory phase, the neutrophil serves little to no beneficial function and further seems to implicate its persistence in a plethora of negative wound healing outcomes. With the data from the Butterworth thesis in mind, our project set out to challenge the negative reputation of the neutrophil and potentially elucidate beneficial functions of this cell persisting well beyond the inflammatory and proliferative phases of healing. As the neutrophil is known to have an enhanced ability to cope and function in a hypoxic environment, we set out to further examine the relationship between the two. Of particular interest were the dynamics of neutrophil persistence, and whether or not they could be related to phenotype, hypoxia and re-vascularization rate.

#### 1.6 Model Validation

While we recognize the current difficulties of generating reliable human data in the field of wound healing, we do believe that any models used must be rigorously validated to ensure adequate recapitulation of the human healing process, if there is any potential therapeutic value to be gleaned from them. Current evidence suggests that healing in mice and humans overlaps sufficiently enough for these animals to be widely used as models of chronic wound healing. While we did not set out with the goal of outright invalidation of this model, we did seek to challenge it with respect to the available data on neutrophil dynamics, and vascular deficiencies widely reported in human and pig studies. Additionally, we sought to validate this model with respect to the expected increased hypoxic load along with impaired re-vascularization which is reported in human studies of diabetic chronic wounds. It must also be noted that while widely

used, this model is not a true representation of chronic wound healing in a sense that all wounds in these animals eventually heal, albeit more slowly.

The finding of similar levels of neutrophil persistence across distinct stages of healing has also been partially validated in a pig model by Dyson et al 1988. This study undertook a quantitative analysis of cell types in the granulation tissue of open pig wounds, with the goal of comparing the effects of different types of bandages. They reported small numbers of neutrophils present at any stage of healing, irrespective of dressing type, and that negligible numbers were observed in wounds after five days. One of the secondary aims of this project was to attempt to confirm this finding with pig wounds from Days 7 and 28.

#### 1.7 The Macrophage

In addition to a multitude of other roles, the macrophage is a critical cell in wounds as it represents the single most effective means of neutrophil clearance <sup>[43]</sup>. Macrophages are capable of responding to neutrophils and their products and are capable of inducing neutrophil apoptosis <sup>[44]</sup>. Perhaps most crucially, these cells are able to recognize and actively ingest apoptotic neutrophils, thus directly helping to resolve wound inflammation <sup>[45–48]</sup>. Several studies suggest that the process of efferocytosis affects macrophage phenotype, causing a switch from pro-inflammatory to a growth enhancing, reparative phenotype <sup>[49]</sup>. Other recent studies suggest that a failure in removal of inflammatory cells, namely neutrophils, plays a role in the pathogenesis of non-healing wounds <sup>[50,51]</sup>. Additionally, a deficit in macrophage ability to effectively remove neutrophils was recently reported to be a critical component of the impaired healing seen in diabetes <sup>[52]</sup>. The investigators implanted sponges into diabetic mice and subsequently isolated the macrophages from these sponges. The isolated macrophages showed a significant impairment in the phagocytosis of apoptotic cells. Interestingly, this

deficit was associated with higher levels of apoptotic cells and pro-inflammatory mediators in wounds, a feature that was further validated in wound tissues of diabetic patients. This data suggests that successful efferocytosis by macrophages may be requisite for appropriate wound healing, both in order to remove apoptotic neutrophils as well as to generate a macrophage phenotype supportive of the proliferative aspects of repair.

#### 1.8 Angiogenesis: What is Considered Normal?

In the unwounded state, endothelial cells in the lumens of blood vessels are in a state of quiescence, evidenced by a basal, balanced expression of pro-anti angiogenic factors in the vascular bed. These factors include vascular endothelial growth factor (VEGF), fibroblast growth factor (FGF), angiopoetin-1 (Ang-1) and pigment epithelium-derived factor (PEDF), and their balanced expression results in a vascular network that is neither proliferating nor diminishing <sup>[53,54]</sup>. Cutaneous injury is accompanied by a sharp increase in local hypoxia, which activates Hypoxia Inducible Factor 1 (HIF-1), which then promotes angiogenesis by upregulating target genes such as VEGF-A. This growth factor stimulates capillaries to form immature loops and branches, a process normally most prominent in the proliferative phase of repair. One of the defining features of normal, early angiogenesis is the formation of disorganized and poorly perfused vasculature, characterized by malformed capillary bed and blind-ended sprouts <sup>[55]</sup>. Although from a structural integrity standpoint these capillaries are sub-optimal, their number in healing skin is much higher than in normal, unwounded skin and peaks at day 7-10 post wounding <sup>[56,57]</sup>.

After the peak of proliferation has passed, a vessel maturation program begins. This process selects for competent nascent vessels to become durable mature vessels similar to the preinjured state. Vessel stabilization is importantly mediated and achieved by pericytes <sup>[58]</sup>. In healing wounds, pericytes are actively recruited in response to several growth factors, with the best described being – platelet derived growth factor (PDGF) <sup>[59,60]</sup>. When these pericytes arrive, they interact with endothelial cells as well as the basement membrane. Capillary pruning is mediated by production of several anti-angiogenic, vascular maturation factors, with the two best studied one's being – pigment epithelium-derived factor (PEDF) and sprouty-2 (SPRY2). PEDF has been shown to be one of the most potent anti-angiogenic factors in the vasculature, and has been demonstrated to be capable of inducing endothelial cell apoptosis as well as reducing the permeability of "leaky" neovasculature <sup>[61]</sup>. In the context of wound healing, studies by Okonkwo et al. have demonstrated its production to be essential for vascular remodeling and maturation <sup>[62,63]</sup>. Sprouty-2 is an intracellular protein that inhibits mitogenactivated protein kinase signaling, ultimately downregulating the effect of VEGF on endothelial cell proliferation in wounds <sup>[63,64]</sup>. An additional level of angiogenic control comes from the interactions of Angiopoetins 1 and 2 (Ang-1/Ang-2) with the Tie2 receptor. Ang-1 is a potent maturation factor and stabilizes pericytes and endothelial cells in capillaries, while Ang-2 has an opposite effect and destabilizes vessels <sup>[65]</sup>. During normal angiogenesis, there are high levels of Ang-2 during the proliferative and pro-angiogenic phases and high levels of Ang-1 during the vessel maturation phase <sup>[66]</sup>. As the remodeling phase of healing draws to an end, the injured tissue achieves normal vascular permeability, blood flow and shows normal vascular branching <sup>[67]</sup>. Perhaps most importantly, the high oxygen demand of the early stages of wound healing, a situation that was previously described to induce and potentially aggravate the activation of many pro-inflammatory mediators and pro-angiogenic factors, returns to preinjury levels <sup>[68]</sup>.

It is worthy of note that angiogenesis overlaps with both the proliferative and remodeling phases of repair <sup>[69]</sup>. Fibroblast migration, proliferation, and collagen synthesis all occur during the same period as the angiogenic response. Recent studies suggest that capillary pruning and collagen maturation may interact in the remodeling phase, with the pruning of capillaries proposed to have a beneficial influence on the final ECM structure <sup>[70]</sup>.

#### 1.9 Diabetic Angiogenesis: What is Wrong?

The previously discussed Ang-1/Ang-2 - Tie2 complex, is one vascular maturation pathway that has been implicated in the angiogenic deficits seen in diabetic wounds. It has been shown that the ratio of Ang-1 to Ang-2 is decreased, strongly suggesting that the ability of diabetic wound neovasculature to undergo a vessel maturation program is likely disturbed <sup>[71,72]</sup>. During the maturation and resolution phases of angiogenesis, PDGF is one growth factor that has been identified as being perturbed in the diabetic state. It promotes capillary maturation by recruiting pericytes and slowing down vessel regression <sup>[73]</sup>. PDGF has been extensively studied in diabetic skin wound healing and db/db mice have been found to express lower levels of PDGF and its receptor in their wounds <sup>[74]</sup>. As previously discussed, neutrophil overabundance often accompanies bouts of repeated, prolonged inflammation. This cell carries elastase, a protease which effectively degrades PDGF, thus likely exacerbating the PDGF deficit already present in the diabetic condition. It is likely that the role of PDGF extends beyond capillary stabilization as it is also a potent mitogen for fibroblasts, thus any perturbations to its normal levels are likely to have profound, widespread effects on different events of the wound healing process, such as re-epithelialization.

A study by Seinz et al. showed that VEGF-A protein and mRNA levels in wounds of db/db mice were significantly decreased, as compared to healthy controls. VEGF-A is a potent proangiogenic factor known for increasing vascular permeability. Follow up studies by Galiano et. Al also reported this deficit, and went on to show that VEGF-A treated mice exhibited more of an early leaky, malformed vasculature and more edema until VEGF-A treatment was ceased <sup>[75]</sup>.

In addition to known diabetic deficits in pro-angiogenic factors, several studies have recently identified deficits in anti-angiogenic and capillary maturation factors in wounds. Production of PEDF was examined in the context of diabetic healing, however this study only looked at serum levels rather than tissue expression levels. One such study reported higher circulating levels of PEDF in patients with a DFU, compared to both healthy and diabetic patients without an accompanying DFU <sup>[76]</sup>. In the context of what PEDF does, this finding is slightly out of line with the others. It must be noted that this study quantified PEDF levels systemically and not within the local wound tissue.

These studies strongly suggest that normal vascular pruning and maturation may be significantly delayed in DFU's, a condition which may greatly contribute to chronicity due to an increased, unresolving hypoxia along with its deleterious pro-inflammatory effects. There is a growing body of evidence that suggest that diabetes results in more tortuous and disorganized vascular architecture. For instance, mice subjected to High Fat Diet induced obesity, a condition meant to closely mimic the diabetic phenotype, present with more tortuous and aberrant vascular network <sup>[77]</sup>. Such findings are paralleled in human diabetic patients, where corrosion casting and scanning electron microscope (SEM) investigations have shown that those suffering from diabetic microangiopathy in the toe region, exhibit damaged capillary architecture and evidence of vascular leakage <sup>[78]</sup>.

MSc. Thesis – April 2024

#### 1.10 Rationale

**Hypothesis:** Neutrophil persistence will result in significantly increased oxygen consumption in the wound bed of the db/db mouse, compared to control. This increased stress will manifest as delayed resolution of hypoxia, delayed wound closure, enhanced neutrophil persistence, and reduced formation of diabetic neo-vasculature.

<u>Aim 1:</u> Investigate the relationship between persistence of neutrophils and delayed hypoxia resolution in the context of impaired diabetic healing.

**Objective:** Confirm that delayed hypoxia resolution correlates with increased persistence of neutrophils in the diabetic wound bed.

<u>Aim 2:</u> Investigate the relationship between hypoxia resolution and the rate of revascularization of tissues.

**Objective:** Confirm that as the wound re-establishes its vascular network, the tissue becomes less hypoxic.

<u>Aim 3:</u> Validate the mouse model as one re-capitulating the major defects seen in impaired human wound healing.

**Objective**: Confirm that the mouse exhibits appropriately comparable wound healing defects to the human.

## Chapter 2

## Characterizing Hypoxia, Neutrophil Persistence and Revascularization in the Murine db/db Model of Type II Diabetic Impaired Skin Healing.

#### Abstract

Typically recruited during early inflammation post-injury, the persistence of neutrophils has been documented in human skin wounds at 4-12 weeks after re-epithelialization is complete. Hypoxia can promote neutrophil survival through an autophagy-like mechanism, with their persistence associated with depletion of molecular oxygen levels in the wound via a process termed "respiratory burst". Depletion of available molecular oxygen in the context of healing can negatively impact a wide array of necessary processes for healing including cell proliferation, collagen deposition and timely resolution of inflammation. Angiogenesis is also a critical milestone during healing as the re-establishment of vascular supply and adequate oxygenation of tissues is required for cutaneous healing. The current project aims at investigating the relationship between hypoxia, neutrophil recruitment to the wound, and angiogenesis in a murine model of diabetic healing.

Based on previously reported data we hypothesized that db/db diabetic mice would exhibit delayed wound closure, increased neutrophil persistence and reduced angiogenesis in comparison with wild-type mice. Selected timepoints for histology were day 3 and day 7, corresponding to the peak of acute inflammation and the peak of proliferation/beginning of angiogenic phase respectively. db/db mice exhibited a significant delay in closure at days 3, 7 and 12 post-wounding. No significant differences were observed in the relative amount of hypoxia in the epithelium at day 7 assessed through HypoxyProbe. A significant increase in *MSc. Thesis – April 2024* 17

the number of hypoxic cells in the granulation tissue of wild-type animals was observed at day 7. Additionally, a significant difference was observed in neutrophil numbers, with increased number in granulation tissue in wounds in db/db vs wild-type at day 7. Finally, no significant difference in the number of CD31<sup>+</sup> cells was observed between phenotypes. No significant differences were observed in the amount of epithelial hypoxia at Day 3. A significant difference was observed in the number of hypoxic cells at Day 3, with wild-type animals showing a higher number in their granulation tissue and one of the edges. There was no difference observed in the number of neutrophils in the wound at Day 3. Lastly, there were more CD31<sup>+</sup> cells in wild-type granulation tissue and at one of the edges.

We conclude that wild-type mice show a minor increase in hypoxia at day 7 post-wounding compared to db/db mice, and db/db mice show a persistence of neutrophils at Day 7. Our current evidence suggests that further validation of the db/db mouse as an impaired wound healing model is needed, as we were not able to reproduce the angiogenic deficit that is often reported in human wounds. The somewhat surprising finding of increased hypoxia in wild-type wounds at days 3 and 7 may further suggest that sustained hypoxia may be a required stimulus for proper wound resolution, however further studies would be required to adequately validate this.

Keywords: hypoxia, inflammation, neovascularization, wound-healing

#### 2.1 Introduction & Significance

Chronic skin wounds are defined as a disrupted or impaired repair process where a sustained anatomical and functional result is not reached within three months <sup>[1]</sup>. Chronic wounds persist as a silent epidemic, adversely impacting the quality of life of over 40 million individuals worldwide <sup>[6]</sup>. Impaired skin wounds fall into three general classifications: venous ulcers, pressure ulcers and diabetic ulcers <sup>[2]</sup>. Venous ulcers affect 1-2% of the adult population, with a higher prevalence in women and the elderly <sup>[3]</sup>, and pressure ulcers are more common in those with compromised mobility or issues with sensory perception; prolonged unrelieved pressure can lead to localized tissue ischemia when applied compression exceeds capillary pressure <sup>[4]</sup>. Diabetes mellitus is a metabolic disease, characterized by elevated blood sugar levels resulting from the body's inability to effectively transport it into cells. There are currently over 382 million people afflicted with this condition worldwide <sup>[7]</sup> and diabetic foot ulcers are a commonly seen and serious complication <sup>[8]</sup>. Approximately 15% of diabetic patients will develop a DFU, with 14-24% subsequently undergoing a lower limb amputation, with a five-year post-op mortality rate from this procedure approaching 50-59% <sup>[9-11]</sup>.

Regardless of their etiology, chronic wounds are all characterized by enhanced local hypoxia and inflammation <sup>[79]</sup>. Hypoxia results from both damage to the vascular structures in the tissue, as well as cell populations that are recruited to the wound site such as neutrophils. Neutrophils have a high oxygen consumption rate <sup>[80]</sup>, but can also persist in hypoxic conditions <sup>[81]</sup>.

Due to these factors, we investigated the dynamics of neutrophil persistence in the db/db model in the context of hypoxia. Primarily, we were interested in seeing if the temporal resolution of wound hypoxia would correlate with a decrease in neutrophil numbers. We were also interested in evaluating the db/db model in the context of data presented by Rosalind Butterworth MD, namely the finding of similar neutrophil numbers observed in the wounds that of varying age, between 4-12 weeks <sup>[82]</sup>. We theorized the db/db model may exhibit a similar neutrophil infiltration pattern to humans, and we investigated whether the db/db injury model possesses parallels with human wound healing. Lastly, we wanted to observe these effects in the context of hypoxia and re-vascularization, as both are known to be able to shape the tissue microenvironment <sup>[83]</sup>

#### 2.2 Materials and Methods

#### 2.2.1 Mice

All animal procedures were carried out in compliance with protocols approved by the University Council on Animal Care at Western University. Wild-type and diabetic (C57BL/6 and BKS.Cg-Dock7m +/+ Leprdb/J) mice were purchased from the Charles River Laboratories (Wilmington, MA). At the time of experiments mice were 10 weeks old. The mice were housed individually and given unrestricted access to food and water. Blood sugar measurements were not taken, however previous work in our lab has reliably demonstrated that mice of the db db phenotype present with significantly higher body weights and had an average blood sugar level of 18.78 mM when compared to wild types who had an average level of 8.59 mM (n=10) <sup>[84]</sup>.

#### 2.2.2 Excisional Wounding Experiments

For all experiments, 4, 4-mm full thickness excisional wounds were created on the dorsal surface in accordance with AUP 2020-142. Pain was controlled for the first 24 hours via intra-peritoneal administration of Buprenorphine at a dose of 17 uL/10 g. Four injections in total were given, 6-8 hours apart. Mice were euthanized with CO<sub>2</sub> and tissue was isolated at Days 1, 3 7 and 12. For assessing closure kinetics, wounds were photographed at Day 0, and then days 1, 3 7 and 12, using a ruler to standardize image dimensions. Image J software was used to calculate wound area.

#### 2.2.3 Human & Porcine Tissue Samples

All human tissues were collected with written, informed consent from patients under protocols approved by the Western University Review Board for Health Sciences Research Involving Human Subjects and in accordance with the 1964 Declaration of Helsinki. Written informed consent was received from participants prior to inclusion and enrolment in the study. Skin samples collected included those from the wound bed and edge, 2 cm proximal to the wound and a non-involved region at least 10 cm away. The samples were then fixed in 4% paraformaldehyde (Sigma Aldrich, St. Louis, MI) overnight and processed to paraffin for histology. Histological analysis of immune cell infiltration was performed on wound and healthy tissue from 5 patients.

Porcine wounding experiments were performed at the West Valley Animal Unit. All procedures were carried out under protocols approved by the Animal Use Subcommittee at Western University. Assessment timepoints were days 3, 7, 14, 28 and 42, however, in this thesis only days 7 and 28 are included. Full-thickness wounds were made bilaterally along the dorso-rostral back skin in each animal, under appropriate anaesthesia. Wounds were made at least 2 cm apart, in order to prevent potential mechanical interference from

neighboring wounds. Upon experimental endpoint the animals were humanely sacrificed, tissue was harvested and fixed in formalin, processed to paraffin, embedded and ultimately sectioned and processed for immunohistochemistry.

#### 2.2.4 Histology

Mouse tissues were fixed in 10% neutral buffered formalin overnight and processed to paraffin. Sections 10 um thick were cut from paraffin blocks and mounted onto FisherBrand<sup>TM</sup> Superfrost Plus microscope slides (Fisher Scientific, Toronto, ON). Attempts were made to keep serially stained sections as close as possible, however, in some cases suboptimal staining required re-runs of stains and additional sectioning. These additional sections were likely some distance away from the originally cut ones. This may account for some of the potential variability seen in wound regions and cell numbers which were analyzed across replicates. Colorimetric staining was performed using the VectaStain HRP -Anti-rabbit detection kit (Vector Laboratories Inc, Burlingame, CA) and a ImmPact DAB Peroxidase Substrate Kit. For Day 3, a total of 8 wild-type and 9 diabetic wounds were analyzed from 3 mice, respectively. For Day 7, a total of 16 wild-type and 14 diabetic wounds were analyzed from 5 mice, respectively. These timepoints were selected for analysis due to their ovelap with the peak of inflammatory phase and beginning of the angiogenic/proliferative phase of healing. Hypoxic cells were identified with the use of Hypoxyprobe Kit (HP1-1000, Burlington, MA). Immune cell infiltration was assessed with an anti-neutrophil elastase rabbit antibody (ab68672) at a dilution of 1:50. Vascular infiltration was assessed with an anti-CD31 rabbit antibody (ab124432) at a dilution of 1:300. For all 3 stains, sections were incubated at 4 degrees Celcius overnight. All stains were run in triplicate.

#### 2.2.5 Cell Counting Quantifications

Hypoxic cell, immune cell and blood vessel infiltration was quantified using QuPATH cellcounting software (Bankhead, P. et al., 2017), as percentage of positive cells per 300x300 um field of view. For counting purposes, two regions of interest were created in the left edge, the wound bed and right edge respectively, totalling six regions per section. The regions of interest were kept in the same locations across replicates and stains. In some cases, the contrast between diaminobenzene (DAB) and hematoxylin was sufficient for the software to recognize positive and negative cells on its own with the default DAB detection vector. However, in other cases where the DAB was clearly present albeit appearing more faintly the software struggled with positively identifying these cells. In these cases, the DAB detection vector was set manually by identifying one clearly labelled well-contrasted cell, drawing a region of interest around it and setting the detection vector to that of the region of interest. This resulted in the software being significantly more accurate in identifying positive cells (Appendix 4.1-4.3), but a balance had to be struck between optimal detection sensitivity and over-sensitivity in order to ensure that only the truly stained cells were being labelled as positive, as opposed to those displaying background DAB signal.

Total cell counts per field of view were also performed using the QuPATH software, in a similar manner described above. Epithelial hypoxia was assessed with the use of Image J software, beginning at the wound edge, and ending at the first obvious drop-off in signal moving laterally away from the wound edge. The scale was manually set for each image by tracing the scale bar. Additionally, sections were sent to Robarts Research to be stained for Picrosirius Red, Van Gieson and Masson's trichrome. There were 7 sections for Day 7, 4 for Day 3 and 2 for Day 1 from both phenotypes respectively.

### 2.2.6 Hypoxyprobe<sup>TM</sup> Assay

Hypoxyprobe (HP1-1000, Burlington, MA) was diluted in solution with 0.9% saline to a final concentration of 6 mg/ml and kept at 4 degrees Celcius in light subdued conditions. At endpoint, Hypoxyprobe solution was administered intra-peritoneally at a dose of 60 mg/kg, 90 minutes prior to euthanasia to allow probe to circulate and bind to targets. Immunohistochemical staining was performed in accordance with the manufacturers protocol, with dilutions of 1:100 used for both primary FITC-MAb1 and secondary rabbit anti-FITC with horseradish peroxidase. Sections were incubated overnight at 4 degrees Celcius with the primary antibody. A logically plausible limitation of this methodology is the destruction of the blood vessels within an intended area of Hypoxyprobe<sup>TM</sup> labeling. Hypoxyprobe<sup>TM</sup> has been reliably demonstrated to label the core of tumors (Appendix 4.9), which are known to possess a disorganized vasculature network <sup>[85]</sup>. These vessels are known to have a diminished ability to deliver nutrients and remove waste <sup>[85]</sup>. However, these tumors still label strongly for Hypoxyprobe<sup>TM</sup>, confirming a delivery mechanism that is able to overcome any present vascular deficiencies.

#### 2.2.7 Statistical Methods

Statistical analyses were performed using the GraphPad Prism version 10.0.1 for Mac, (GraphPad Software, Boston, MA). Graphs were also generated with the Prism software. Normality testing was carried out, with nearly all data sets showing a non-parametric distribution. Because of this, a non-parametric Mann-Whitney t-test was used and values of P < 0.05 were taken to be statistically significant.
#### 2.3 Results

# **2.3.1** Diabetic mice exhibit a delay in the rate of wound contraction, re-epithelialization and collagen deposition.

Differences in wound closure rates between wild-type and db/db diabetic mice have been well described in the literature, including by our group. In the first experiments, we confirmed whether closure kinetics, re-epithelialization and granulation tissue formation in wild-type and diabetic mice differed. A significant reduction in wound closure was observed in diabetic mice at days 3, 7 and 12 in comparison with wild-type mice (Figure 1). Diabetic mice exhibited reduced re-epithelialization at day 7 compared to wounds in wild type mice (Figure 2). No qualitative differences in collagen deposition were observed at days 1 and 3 between phenotypes (Figure 3), although wounds in db/db mice exhibited less collagen deposition at day 7 versus their wild-type counterparts (Figure 4). **Figure 1** - Diabetic mice exhibit a significant delay in closure at days 3, 7 and 12. (A) – Visual comparison of wounds across timepoints. (B) – Trend graph of percent closure differences between phenotypes. Significant differences were observed at Day 3 (p=0.015), 7 (p=<0.0001) and 12 (p=0.0104); (p < 0.05, Mann-Whitney t-test). (C) – Analysis of wound closure percentage relative to Day 0. Calculation performed as [measured wound area] / [original wound area]x100.





**Figure 2** – Masson's Trichrome Stain showing a failure of re-epithelialization at Day 7 in diabetic animals, with a clear break still present in epithelium. Arrows denoting wound margins.



## Masson's Trichrome - Days 1, 3 and 7

**Figure 3** – Picrosirius Red Stain showing no visual differences in the amount of collagen deposition between Days 1 (A) and 3 (B). No striking differences in wound morphology. Arrows denoting wound margins.





**Figure 4** – Picrosirius Red staining showing more collagen in the wound beds of wild-type animals at Day 7. Re-epithelialization complete at Day 7 in wild-type animals. Arrows denoting wound margins.



### Picrosirius Red - Polarized and Bright Field - Day 7

#### 2.3.2 – Wild-type wounds are qualitatively more hypoxic at day 7 post-wounding.

Upon full-thickness dermal injury, it is thought that the wound bed is hypoxic due to the destruction of blood vessels within it. To assess the level of cellular hypoxic load present in wounds, tissues were labeled with Hypoxyprobe<sup>TM</sup>. No major differences in wound morphology were observed between phenotypes (Figure 5). Wild-type wound beds qualitatively appeared more hypoxic at Day 7, with prominent epithelial staining. Note DAB staining in hair follicles as a positive control. A "wave" of hypoxic cells could be observed at the wound edges, correlating with profound cell infiltration at Day 7. Cells in uninjured dermis were less densely spread, yet still often stained hypoxic.

**Figure 5** – General appearance of wild-type and diabetic wounds processed with Hypoxyprobe<sup>TM</sup>. No striking differences in hypoxic staining localization or wound morphology. Wild-type wound beds qualitatively appear more hypoxic at Day 7, with prominent epithelial staining evident above the wound.



Hypoxyprobe Days 1, 3 and 7

## **2.3.3** There are no significant differences in the amount and persistence of epithelial hypoxia between phenotypes at Days 3 & 7.

Prior to undertaking the quantitative assessment of the amount of hypoxic cell infiltrate in wounds, we first assessed the persistence of hypoxia in the epithelium of the wounds, moving bi-laterally away from the wound bed. A sharp increase in hypoxic signal was observed at the immediate edges of the wound in both phenotypes, which would then gradually decrease with increasing distance from the wound bed. There were no significant differences in the amount of epithelial hypoxia at either Day 3 or 7 (Figure 6, Figure 7).

Day 3 biological replicates WT: 3, technical replicates WT: 24.

Day 3 biological replicates db/db: 3, technical replicates: 26.

Day 7 biological replicates WT: 5, technical replicates WT: 48.

Day 7 biological replicates db/db: 5, technical replicates db/db: 41.

**Figure 6a -** Hypoxic signal goes up sharply in the epithelium directly proximal to the wound bed and then drops off. There were no significant differences in the length of persistence of hypoxic signal in the epithelium between phenotypes at Day 3.

Epithelial Hypoxia Distance - Day 3



**Figure 6b** – Hypoxic signal goes up sharply in the epithelium directly proximal to the wound bed and then drops off. There were no significant differences in the length of persistence of epithelial hypoxic signal between phenotypes at Day 7.



**Figure 6c** - Epithelial distance ( $\mu$ M) in which hypoxic signal was evident. No significant differences were observed in the amount of hypoxic signal persistence in the epithelium at Day 3 or 7 between phenotypes (p<0.05, Mann-Whitney t-test).



Epithelial Hypoxia - Day 3

Epithelial Hypoxia - Day 7



6c)

## 2.3.4 Wild-type wounds present with more hypoxic cells in their wound beds and edge at Day 3 and more hypoxic cells in wound beds at Day 7.

We assessed the number of hypoxic cells in the wound beds and edges of wild-type and diabetic animals. At day 3, wild-type mice had significantly more hypoxic cells at one of the edges, and wound beds (Figure 8b). At day 7, there were significantly more hypoxic cells in the wound beds of wild-type animals. The edges were not significantly different. (Figure 9b).

Day 3 biological replicates WT: 3, technical replicates WT: 24.

Day 3 biological replicates db/db: 3, technical replicates db/db: 26.

Day 7 biological replicates WT: 5, technical replicates WT 48.

Day 7 biological replicates db/db: 5, technical replicates db/db: 41.

**Figure 7a** – Wild-type mice exhibited a higher hypoxic cell load at Day 3 at the right edge and wound bed.





7a)

**Figure 7b** – Wild-type mice exhibited a higher hypoxic cell load at Day 7 in the wound beds. There were no significant differences in hypoxic cell load present at the edges in either phenotype.





**Figure 7c** – Different wound regions exhibited distinct profiles in regard to hypoxic cell load at Day 3. There was a significantly higher percentage of hypoxic cells at the wild-type right edges (p=0.0179, Mann Whitney t-test) and wound beds (p=0.0048, Mann Whitney t-test) per field of view assessed. At day 7 wild-type wounds exhibited a significantly higher percentage of hypoxic cells in their wound beds (p=0.0238, Mann Whitney t-test) per field of view assessed.



7c)

**2.3.5** There are no significant differences in neutrophil load at Day 3. Diabetic wounds present with a higher neutrophil load in their wound beds at Day 7.

Due to their enhanced ability to cope with hypoxic environments, as well as the widely reported dysfunction in neutrophil clearance in the context of diabetic healing, next we assessed the differences in neutrophil load between phenotypes in the context of a hypoxic environment. At day 3, there were no significant differences in neutrophil load (Figure 10b), however, wild-type wounds did present with a slightly elevated neutrophil count, although this was not statistically significant.

At day 7, a significantly elevated neutrophil count was observed in the wound beds of diabetic mice (Figure 11b). This finding was in line with the majority of reported data in the field, which tends to describe a neutrophil overabundance in diabetic wound beds along with a number of deleterious effects their prolonged presence causes.

Day 3 biological replicates WT: 3, technical replicates WT: 24.

Day 3 biological replicates db/db: 3, technical replicates db/db: 26.

Day 7 biological replicates WT: 5, technical replicates WT: 48.

Day 7 biological replicates db/db: 5, technical replicates db/db: 41.

**Figure 8a** – There were no significant differences between phenotypes in neutrophil load at Day 3.

Neutrophil Elastase Count - Day 3



Diabetic



**Figure 8b** – Diabetic mice exhibited a higher neutrophil load in their wound beds at Day 7.

Day 7 - Neutrophil Count



**Figure 8c** – No significant differences in percentage of neutrophil elastase labelled cells per field of view at Day 3. Wild-type mice exhibited a slightly higher percentage of positive cells at their wound beds but this was not statistically significant (p<0.05, Mann-Whitney t-test). At Day 7 Diabetic wounds exhibited a significantly higher percentage of neutrophil elastase labelled cells per field of view assessed at Day 7 (p=0.0308, Mann Whitney t-test).

Neutrophil Count Day 3



8c)

MSc. Thesis – April 2024

# 2.3.6 Wild-type mice exhibited significantly more CD31<sup>+</sup> cells at Day 3 in their wound beds and right edges. There were no significant differences in amount of CD31<sup>+</sup> blood vessels at Day 7 between phenotypes.

As we were interested in the relationship between hypoxia, neutrophil infiltration and new blood vessel formation, the next population of cells we assessed was CD31<sup>+</sup>. At day 3, there were significantly more CD31<sup>+</sup> cells in the wound beds and right edges of wild-type wounds (Figure 12b). There were no significant differences in the amount of CD31<sup>+</sup> cells between phenotypes at Day 7 (Figure 13b).

Day 3 biological replicates WT: 3, technical replicates WT: 24.

Day 3 biological replicates db/db: 3, technical replicates db/db: 26.

Day 7 biological replicates WT: 5, technical replicates WT: 48.

Day 7 biological replicates db/db: 5, technical replicates db/db: 41.

**Figure 9a** – Wild-type mice exhibited a significantly higher number of  $CD31^+$  cells in their wound beds and right edge.

Day 3 - CD31


**Figure 9b**) – There were no significant differences in the number of  $CD31^+$  cells at any of the wound locations assessed at Day 7.

Day 7 - CD31



**Figure 9c** – Wild-type wounds exhibited a significantly higher percentage of CD31<sup>+</sup> cells per field of view assessed at Day 3 in their wound beds (p=0.0107, Mann Whitney t-test) and right edges (p=0.0006, Mann Whitney t-test). At Day 7 there were no significant differences in the percentage of CD31<sup>+</sup> cells per field of view assessed (p<0.05, Mann Whitney t-test).



9c)

# 2.3.7 Neutrophil elastase staining in both human and porcine tissue.

As neutrophils comprised one of the core focus areas of this project, and one of the central aims was evaluation of the degree of correspondence between humans and mice, the following figures (Figure 10 and 11) show the neutrophil elastase staining pattern in both humans and pigs. Note in porcine tissue there is a decrease in neutrophils between Days 7 and 28.

**Figure 10** – Human chronic wounds present with an abundance of neutrophils at their wound edges and underneath the eschar.



**Figure 11** – Porcine wounds present with an abundance of neutrophils at their wound edges at Day 7. There is a qualitative decrease in neutrophil number evident at Day 28.



11)

# **2.4 Discussion**

Following full thickness injury to skin, hypoxia increases in the damaged tissue which initiates a cascade of inflammatory and angiogenic cellular processes to facilitate tissue repair. Neutrophils which are initially recruited and are present in hypoxic conditions have been shown in tissues such as the intestine to induce hypoxic signaling at the transcriptional level in epithelial cells <sup>[80]</sup>. Furthermore, activated neutrophils at sites of inflammation can further increase tissue hypoxia due to their high oxygen consumption, a process termed inflammatory hypoxia <sup>[86]</sup>. It was shown in 1991 by Rosalind Butterworth that neutrophils persist in the healing dermis of human skin wounds even after re-epithelialization is complete, but the underlying reason is not known. Associated with and known to cause hypoxia, studying neutrophils and their relationship to oxygen levels in humans is problematic due to ethical considerations and the inherent variability between cellular and tissue level hypoxia. The aim of this study was to investigate on the cellular level the relationship between hypoxia, neutrophil persistence and re-vascularization using a murine diabetic model of delayed skin wound healing.

We first focused on quantification of hypoxic signal in cells at the wound edge as well as in the developing granulation tissue. To our knowledge, this is one of the first studies utilizing Hypoxyprobe<sup>TM</sup> to assess hypoxia in a full-thickness excisional murine skin healing. Previous studies have used Hypoxyprobe<sup>TM</sup> in analysis of endometrial shedding, where they demonstrated a temporal gradient of hypoxia arising as the endometrial vessels regressed and the lining was shed <sup>[87]</sup>. In the context of tissue repair, work by Kang et al. demonstrated that following a gastrocnemius or hindpaw incision at days 1, 2, 4 and 10, a temporal hypoxic gradient develops. Staining was most intense at days 1 and 2, and by day 4 staining had become

diffuse and little staining was reported by day 10<sup>[88]</sup>. In our study, we showed that wounds in both WT and db/db mice showed Hypoxyprobe<sup>TM</sup> signal localized strongly in the epithelial edges directly proximal to the wound bed. Hair follicles in both phenotypes uniformly stained positive, providing a positive control to our assay as the cells of hair follicles are known to be hypoxic<sup>[89]</sup>. Tracing the signal from the wound edge laterally using Image J until the first obvious drop in signal intensity showed no significant differences in epithelial hypoxia between phenotypes at either day 3 or 7. In a subset of wounds irrespective of phenotype, a hypoxic "trail" was observed at the end of the epithelium, extending into the wound bed under the eschar (See Figure 7a). The presence of hypoxia in the epithelium is not unexpected as it is avascular and damage to the skin removes the vascular supply <sup>[90]</sup>. No significant differences in epithelial hypoxia were observed between phenotypes at either day 3 or 7, with both phenotypes exhibiting an increase in the number of hypoxic epithelial cells between days 3 and 7. This effect was larger in diabetic mice, which had a lower number of hypoxic cells at day 3. Interestingly, the significantly increased hypoxic epithelium at day 3 and 7 observed in the wild-type mice, correlated with significantly faster wound closure kinetics at days 3, 7 and 12 (Figure 1), increased re-epithelialization at day 7 (Figure 2) and increased wound bed collagen deposition at day 7 (Figure 4). We hypothesize the defects in these healing parameters in the db/db mouse to be a possible manifestation of insufficient hypoxia adaptation in the db/db mouse, more specifically an insufficient recruitment of early reparative cell populations because of a failure to respond and adapt to hypoxia.

Of great significance for wound healing and impaired healing models, we demonstrate for the first time that wild-type mice have significantly higher total number of hypoxic cells at day 3 at the wound edges and wound bed as well as in the day 7 wound bed compared to db/db mice.

As previous research highlighted a reduction in cell infiltration in db/db mice<sup>[91]</sup>, we confirmed that the increase in hypoxic cells in WT mice wounds in comparison to db/db wounds was not due to differences in total cell number (Appendix 4.4-4.5). Although wounds in wild-type mice trended towards a higher total average cell infiltrate compared to db/db wounds, it was not significant. Our data suggests that WT wounds exhibit a greater degree of adaptation and responsiveness to hypoxia, compared to db/db mice, casting doubt on the db/db mouse model representing delayed healing due to increased hypoxia. This data is consistent with previous work in the field showing less micron RNA-210 (miR-210) expression in db/db wounds as compared to wild-type. miRNAs are a group of tightly conserved non-coding RNAs, and their dysregulated expression has been shown to have the potential to drastically alter translation of multiple genes in a cell <sup>[92]</sup>. miR-210 in particular, has been shown to be the highest miRNA upregulated in hypoxic cells, and is also known to upstream of the regulation of a number of genes involved in energy metabolism and tissue repair <sup>[93]</sup>. Narayanan et al., demonstrated that induction of miR-210 is significantly downregulated in the wounds of diabetic patients and db/db mice, and further demonstrated this effect to be dose dependent due to hyperglycemia. When mice were treated with an miR-210 clone, inflammation was reduced yet granulation tissue formation, proliferation and angiogenesis all improved. As such, they attributed the defects observed in the db/db mouse to miR-210 insufficiency. Further work by Catarina et al., demonstrated a failure of stabilization of a critical hypoxia response regulator transcription factor HIF-1 $\alpha$ , and linked this defect to hyperglycemia <sup>[94]</sup>. Our data is in agreement with current evidence in the field, namely that db/db animals upon injury to the skin exhibit a delay in adaptation to hypoxia resulting in delayed wound healing outcomes.

In both phenotypes, at day 3, most cells infiltrating the granulation tissue at the edges were positive for Hypoxyprobe<sup>TM</sup>, suggesting that these wound regions were likely perceived as

more hypoxic by the cells in comparison to the wound bed. This effect became even more pronounced at day 7, as the overall number of infiltrating cells increased significantly in both phenotypes, with most cells at the wound edges and the wound bed being hypoxic. We observed a clear "volcano shape" of hypoxic cells infiltrating from deeper in the tissue under both sides of the wound margins (Appendix 4.6). It is noteworthy that although there were far less cells in uninjured tissue, many of them still stained positive for Hypoxyprobe<sup>TM</sup> especially deeper in the reticular dermis, hypodermis and areas around the panniculus carnosus muscle. This suggests that damage to the skin in one area may impact on the vascular trees in adjacent regions that would be seen as "intact" or injured skin with hypoxia occurring.

The finding of increased hypoxic sensing in wild-type mice, seemed somewhat counterintuitive at first, as we initially hypothesized that due to the widely reported angiogenic deficiencies and an exaggerated inflammatory response <sup>[95,96]</sup>, we would observe an increased hypoxic load in the diabetic wounds. Although counter to our initial hypothesis, our observations could plausibly be explained by a more metabolically active and thus energy demanding environment in the wild-type wounds, potentially accounting for better healing outcomes i.e., increased closure kinetics and collagen deposition seen in this population. Our evidence suggests that the sharp, early increase of acute hypoxia may be a positive prognostic sign for normal healing dynamics, as it's a well-recognized critical regulator of early recruitment of multiple populations of reparative cells <sup>[79]</sup>. Further research is needed to better understand at what point acute, beneficial hypoxia becomes chronic in skin healing, and what's perceived as acute versus chronic on a cellular level. Due to their ability to function in hypoxic microenvironments we next assessed the level of neutrophil infiltrate in wounds at day 3 and day 7, and whether their persistence correlated with the presence of hypoxia in the surrounding tissue and wound bed. Overall, in both phenotypes at both day 3 and day 7, a few neutrophils were observed in the papillary dermis, with most infiltrating neutrophils emerging from below the panniculus carnosus muscle. Day 3 and 7 wounds in both WT and db/db mice exhibited a pronounced staining pattern directly under the base of the eschar, which in a subset of wounds extended the entire length of the wound. There were no significant differences observed in neutrophil numbers at day 3 between phenotypes, however wild-type mice had slightly higher counts in their wound beds and at one of the edges. In general, the middle core of the wound bed exhibited a lower neutrophil load compared to the superficial portion, the wound edges and the deep portion. The finding that neutrophils are invading from the base of the wound as opposed to the edges bears significance in the context of Hartwell's work, which showed that the wound edges and upper dermis immediately adjacent to the wound are quite inert and do not participate in any significant way in the reparative process <sup>[97]</sup>. This finding has significance for any biomaterial wound treatment therapies, and suggests that any exogenous materials placed in the wound may have greater success if placed deeper in the healing tissue as opposed to superficial placement.

We report a finding of significantly elevated neutrophil numbers in the day 7 wound beds in db/db mice. These findings are consistent with the widely reported increased neutrophil persistence in both murine and human diabetic healing literature <sup>[98]</sup>. As expected, in contrast to db/db wounds, we observed a reduction in neutrophil numbers in the wild-type mice from day 3 to day 7. This finding is in line with previous studies showing that neutrophil presence and activity peak between days 3-5 of the late inflammatory phase of healing, and subsequently

drop as the wound moves to the proliferative phase <sup>[99]</sup>. Our data suggests that the mouse does not show a similar neutrophil count pattern as described in humans in wound bed tissue regardless of the timepoint assessed. Our data further suggests a potential deficit in early hypoxia adaptation in the db/db mouse, as hypoxia is a known powerful mediator of early neutrophil recruitment and energy metabolism <sup>[100]</sup>. We did not observe a significant difference in neutrophil number until day 7 in the db/db, suggesting that the entire wound healing process in the db/db mouse is delayed with significant neutrophil presence manifesting at day 7 by which time it is generally postulated that their numbers should be decreasing. If a defect in hypoxia adaptation exists in the db/db mouse, and hypoxia is a known regulator of neutrophil recruitment and energy metabolism, we postulate that the significantly elevated neutrophil numbers at day 7 in the db/db mouse. This data parallels findings in the human, which show that inflammation persists or occurs slower in the wounds of diabetic patients <sup>[101]</sup>.

The previously discussed defect in HIF-1 $\alpha$  stability in db/db mice during wound healing described by Catarina et al., bears implications for angiogenesis as well. HIF-1 $\alpha$  binds to the hypoxia response element (HRE) in the gene promoter region of vascular endothelial factor (VEGF), which in turn stimulates angiogenesis <sup>[79]</sup>. Thus, if a defect exists in HIF-1 $\alpha$  stability in the diabetic, a potential manifestation of this could be a delayed recruitment of necessary amounts of endothelial progenitor cells. Immediately following any insult resulting in full thickness dermal injury, blood vessels are damaged which initiates a cascade of reparative events, along with a sharp rise in local tissue hypoxia <sup>[79]</sup>. It is known that hypoxia serves among several other factors to increase recruitment of many different cell populations

including neutrophils, fibroblasts, and endothelial progenitor cells to the site of injury <sup>[79]</sup>. Previous work has shown that hypoxia is a necessary physiological event for early repair to proceed successfully <sup>[79]</sup>. As hypoxia arises from local disturbance and damage to the microcirculation, we next quantified vascular density and blood vessels infiltration of the granulation tissue using CD31 as a marker. In general, we did not observe any major differences in CD31 localization between phenotypes at either day 3 or 7. In both phenotypes, extensive vasculature was observed in the panniculus carnosus layer, and the layer just above it that resembled fatty pockets. Most positively stained rounded and elongated cells, presumed to be monocytes or endothelial progenitors, seemed to come from below the panniculus carnosus, much deeper in the tissue, which has been previously identified in humans as an area from which cells are recruited <sup>[97]</sup>. At day 3, increased endothelial progenitor populations in the wound beds was observed, although endothelial tubes with lumens were also present. At day 7, we observed very dense vasculature with clear lumens at both the immediate edges and the granulation tissue itself, consistent with previous work demonstrating extensive vascularization during healing in the human. Work by DiPietro et al., showed that as adult human skin wounds heal, a period of rapid capillary growth occurs with many more capillaries forming in the wound bad than are normally present in uninjured tissue <sup>[102]</sup>. This group also showed that as healing progresses, most of these newly formed capillaries regress and the terminal vascular density becomes similar to that of uninjured tissue. Our data was in line with this work, showing a lot more CD31<sup>+</sup> vessels in the granulation tissue than in uninjured skin. These vessels also appeared to contour and align parallel to the wound edges.

Although the localization pattern of the stain did not appear dramatically different between phenotypes, we report a significant difference in the amount of CD31<sup>+</sup> cells in the wild-type

right edge and wound bed at day 3. Although this timepoint is well before the defined angiogenic peak, this data could be explained when considered together with the evidence obtained from the Hypoxyprobe<sup>TM</sup> assay; at day 3 we observed more hypoxic cells at both the wild-type wound edge and the right edge. Hypoxia is a powerful initiator of angiogenesis, via the transcription factor HIF-1 $\alpha$ <sup>[79]</sup>. When activated, it binds to the gene promoter region of the VEGF gene, which in turn upregulates VEGF<sup>[79]</sup>. This factor is a major angiogenic factor which stimulates endothelial progenitor cells to migrate and form capillaries <sup>[79]</sup>. Thus, it is possible that the higher level of CD31<sup>+</sup> cells could be explained by a more elevated level of local cellular hypoxia. This postulation is made stronger by the fact that the significant difference in degree of vascular infiltration at day 3 (Figure 9c), was coupled with a significant difference in hypoxic cell load in the same wound locations at the same timepoint (Figure 7c). No significant differences were observed between phenotypes at day 7. This could suggest that if an angiogenic deficit existed at day 3, the db/db mouse may have been able to compensate or recover angiogenic processes by day 7. Our current data suggests that the db/db impaired healing model does not consistently re-capitulate the angiogenic deficit commonly reported in the diabetic healing literature <sup>[95]</sup>. Recent work from our lab suggests that it is not a failure of angiogenesis to occur, but rather a premature stabilization of the developing vessels by pericytes and mural cells <sup>[103]</sup>. Other work in the field has outlined the existence of a vascular "plasticity window", during which newly formed vessels mature and become remodeled <sup>[104]</sup>. This work further postulates that it is the acquisition of a vascular pericyte coating that marks the end of this plasticity window. We postulate that it is not necessarily the number of new blood vessels in the granulation tissue that constitutes successful angiogenesis, but the amount of these vessels that undergo further remodeling and maturation that accounts for physiologically desirable wound healing outcomes. If the vessels in the db/db mouse become prematurely wrapped, and are thus unable to undergo any further remodeling, this can plausibly create issues with their future functionality and levels of perfusion. Interestingly, the significance co-relation observed at day 3 between increased hypoxia and increased vascularization, was absent at day 7. Wild-type wounds presented with more hypoxia, but this was not correlated with an increased number of CD31<sup>+</sup> cells. This data is suggestive of an earlier window for the beneficial effects of hypoxia in stimulating angiogenesis and optimal tissue repair.

Overall, our data supports the theory of the importance of adaptation to early hypoxia in the context of wound healing previously outlined by other research groups <sup>[105]</sup>. Hypoxia and its insufficient adaptation by reparative cells can logically be implicated in all three areas of interest in our study – perceived hypoxic load in the wound and adaptation to it by resident cells, inflammation and neutrophil recruitment, and angiogenesis <sup>[94][100][106]</sup>. We postulate that the data gleaned from the db/db model in this present study, suggests not a failure of wound healing processes, but a delayed manifestation, as all the physiologically required processes for wound healing were observed in the db/db mouse, albeit with a significant delay. We suggest that further work should focus on the temporal induction of HIF-1 $\alpha$  in the earliest stages of healing, ideally, in aged mouse models as most chronic wounds occur in this population <sup>[107]</sup>. The mice used in the present study were approximately 5 months old. According to Jackson Laboratories, mice ranging from 18-24 months correlate with humans ranging from 56-69 years of age. This fact brings forth an inherent limitation of our study, namely the age of the mice utilized being 5 months, thus likely limiting the translation of our present data to delayed wound healing in the younger human. Taken altogether, in our view future chronic wound healing studies should aim to utilize aged db/db mice at least 1-1.5 years

old, within that cohort we suggest an insulin treatment group in order to further elucidate the effect of hyperglycemia on cellular hypoxic adaptation, HIF-1 $\alpha$  stability, and its subsequent activation in various reparative cell populations.

### 2.5 References

- [1] R. Nunan, K. G. Harding, P. Martin, *DMM Disease Models and Mechanisms* **2014**, *7*, 1205.
- [2] T. Mustoe, The American Journal of Surgery 2004, 187, S65.
- [3] M. A. Fonder, G. S. Lazarus, D. A. Cowan, B. Aronson-Cook, A. R. Kohli, A. J. Mamelak, *J Am Acad Dermatol* **2008**, *58*, 185.
- [4] T. Hurd, *Wounds International* **2013**, *4*, 13.
- [5] P. Driscoll, "Wound prevalence and wound management," can be found under http://blog.mediligence.com/2013/01/29/wound-prevalence-and-wound-management-2012-2020., 2012.
- [6] C. S. Garwood, J. S. Steinberg, P. J. Kim, *Clin Podiatr Med Surg* 2015, 32, 121.
- [7] D. Baltzis, I. Eleftheriadou, A. Veves, *Adv Ther* **2014**, *31*, 817.
- [8] A. Alavi, R. G. Sibbald, D. Mayer, L. Goodman, M. Botros, D. G. Armstrong, K. Woo, T. Boeni,
  E. A. Ayello, R. S. Kirsner, J Am Acad Dermatol 2014, 70, 1.e1.
- [9] W. J. Jeffcoate, K. G. Harding, *Lancet* **2003**, *361*, 1545.
- [10] D. Martins-Mendes, M. Monteiro-Soares, E. J. Boyko, M. Ribeiro, P. Barata, J. Lima, R. Soares, J Diabetes Complications 2014, 28, 632.
- [11] S. C. Wu, V. R. Driver, J. S. Wrobel, D. G. Armstrong, *Foot Ulcers in the Diabetic Patient, Prevention and Treatment*, **2007**.
- [12] V. Falanga, *The Lancet* **2005**, *366*, 1736.
- [13] L. Prompers, N. Schaper, J. Apelqvist, M. Edmonds, E. Jude, D. Mauricio, L. Uccioli, V.
  Urbancic, K. Bakker, P. Holstein, A. Jirkovska, A. Piaggesi, G. Ragnarson-Tennvall, H. Reike, M.
  Spraul, K. Van Acker, J. Van Baal, F. Van Merode, I. Ferreira, M. Huijberts, *Diabetologia* 2008, 51, 747.
- [14] M. A. Fonder, G. S. Lazarus, D. A. Cowan, B. Aronson-Cook, A. R. Kohli, A. J. Mamelak, J Am Acad Dermatol 2008, 58, 185.
- [15] T. Mustoe, *The American Journal of Surgery* **2004**, *187*, S65.
- [16] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [17] T. Mustoe, *The American Journal of Surgery* **2004**, *187*, S65.
- [18] R. Diegelmann, M. Evans, Frontier Biosciences 2004, 283.
- [19] R. Goldman, Adv Skin Wound Care 2004, 17, 24.

- [20] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [21] J. Li, J. Chen, R. Kirsner, *Clin Dermatol* **2007**, *25*, 9.
- [22] R. F. Diegelmann, M. C. Evans, Front Biosci 2004, 9, 283.
- [23] S. Nagaraja, A. Wallqvist, J. Reifman, A. Y. Mitrophanov, J Immunol 2014, 192, 1824.
- [24] D. Baltzis, I. Eleftheriadou, A. Veves, *Adv Ther* **2014**, *31*, 817.
- [25] R. Blakytny, E. Jude, *Diabet Med* **2006**, *23*, 594.
- [26] M. Peppa, P. Stavroulakis, S. A. Raptis, Wound Repair Regen 2009, 17, 461.
- [27] R. Gary Sibbald, K. Y. Woo, *Diabetes Metab Res Rev* **2008**, *24 Suppl* 1, S25.
- [28] J. Berlanga-Acosta, G. S. Schultz, E. López-Mola, G. Guillen-Nieto, M. García-Siverio, L. Herrera-Martínez, *Biomed Res Int* 2013, 2013, 1.
- [29] R. F. Diegelmann, M. C. Evans, Front Biosci 2004, 9, 283.
- [30] R. Zhao, H. Liang, E. Clarke, C. Jackson, M. Xue, Int J Mol Sci 2016, 17, DOI 10.3390/ijms17122085.
- [31] L. H. Toledo-Pereyra, F. Lopez-Neblina, A. H. Toledo, Ann Transplant 2004, 9, 81.
- [32] H. K. Eltzschig, P. Carmeliet, N Engl J Med 2011, 364, 656.
- [33] W. Ertel, M. H. Morrison, A. Ayala, I. H. Chaudry, Am J Physiol 1995, 269, R160.
- [34] G. Hartmann, M. Tschöp, R. Fischer, C. Bidlingmaier, R. Riepl, K. Tschöp, H. Hautmann, S. Endres, M. Toepfer, *Cytokine* **2000**, *12*, 246.
- [35] C. Brigati, B. Banelli, A. di Vinci, I. Casciano, G. Allemanni, A. Forlani, L. Borzì, M. Romani, Mediators Inflamm 2010, 2010, 263914.
- [36] H. Rosen, J. R. Crowley, J. W. Heinecke, *J Biol Chem* **2002**, *277*, 30463.
- [37] C. W. Baldridge, R. W. Gerard, *American Journal of Physiology-Legacy Content* **1932**, *103*, 235.
- [38] J. Karhausen, V. H. Haase, S. P. Colgan, Cell Cycle 2005, 4, 256.
- [39] A. F. McGettrick, L. A. J. O'Neill, *Cell Metab* **2020**, *32*, 524.
- [40] E. Kolaczkowska, P. Kubes, Nat Rev Immunol 2013, 13, 159.
- S. R. Walmsley, C. Print, N. Farahi, C. Peyssonnaux, R. S. Johnson, T. Cramer, A. Sobolewski, A.
  M. Condliffe, A. S. Cowburn, N. Johnson, E. R. Chilvers, *J Exp Med* 2005, *201*, 105.
- [42] T. N. Demidova-Rice, M. R. Hamblin, I. M. Herman, Adv Skin Wound Care 2012, 25, 304.
- [43] T. J. Koh, L. A. DiPietro, *Expert Rev Mol Med* **2011**, *13*, e23.
- [44] A. J. Meszaros, J. S. Reichner, J. E. Albina, *J Immunol* **2000**, *165*, 435.

- [45] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [46] A. J. Meszaros, J. S. Reichner, J. E. Albina, *J Leukoc Biol* **1999**, *65*, 35.
- [47] J. M. Daley, J. S. Reichner, E. J. Mahoney, L. Manfield, W. L. Henry, B. Mastrofrancesco, J. E. Albina, J Immunol 2005, 174, 2265.
- [48] J. S. Savill, A. H. Wyllie, J. E. Henson, M. J. Walport, P. M. Henson, C. Haslett, J Clin Invest 1989, 83, 865.
- [49] V. A. Fadok, D. L. Bratton, A. Konowal, P. W. Freed, J. Y. Westcott, P. M. Henson, J Clin Invest 1998, 101, 890.
- [50] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [51] J. B. Acosta, D. G. del Barco, D. C. Vera, W. Savigne, P. Lopez-Saura, G. Guillen Nieto, G. S. Schultz, *Int Wound J* **2008**, *5*, 530.
- [52] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [53] T. N. Demidova-Rice, J. T. Durham, I. M. Herman, Adv Wound Care (New Rochelle) 2012, 1, 17.
- [54] M. G. Tonnesen, X. Feng, R. A. Clark, J Investig Dermatol Symp Proc 2000, 5, 40.
- [55] D. Fukumura, R. K. Jain, *APMIS* **2008**, *116*, 695.
- [56] K. E. Johnson, T. A. Wilgus, Adv Wound Care (New Rochelle) 2014, 3, 647.
- [57] A. M. Szpaderska, J. D. Zuckerman, L. A. DiPietro, J Dent Res 2003, 82, 621.
- [58] K. K. Hirschi, P. A. D'Amore, *Cardiovasc Res* **1996**, *32*, 687.
- [59] G. Allt, J. G. Lawrenson, *Cells Tissues Organs* **2001**, *169*, 1.
- [60] P. Lindblom, H. Gerhardt, S. Liebner, A. Abramsson, M. Enge, M. Hellstrom, G. Backstrom, S. Fredriksson, U. Landegren, H. C. Nystrom, G. Bergstrom, E. Dejana, A. Ostman, P. Lindahl, C. Betsholtz, *Genes Dev* 2003, 17, 1835.
- [61] M. L. Broadhead, S. P. Becerra, P. F. M. Choong, C. R. Dass, *Growth Factors* 2010, 28, 280.
- [62] M. S. Wietecha, M. J. Król, E. R. Michalczyk, L. Chen, P. G. Gettins, L. A. Dipietro, *Am J Physiol Heart Circ Physiol* **2015**, *309*, H812.
- [63] M. S. Wietecha, M. J. Król, E. R. Michalczyk, L. Chen, P. G. Gettins, L. A. DiPietro, *Am J Physiol Heart Circ Physiol* **2015**, *309*, H812.
- [64] M. S. Wietecha, W. L. Cerny, L. A. DiPietro, *Curr Top Microbiol Immunol* **2013**, *367*, 3.
- [65] C. A. Staton, M. Valluru, L. Hoh, M. W. R. Reed, N. J. Brown, *Br J Dermatol* **2010**, *163*, 920.

- [66] Y. Brudno, A. B. Ennett-Shepard, R. R. Chen, M. Aizenberg, D. J. Mooney, *Biomaterials* 2013, 34, 9201.
- [67] P. Carmeliet, *Nature* **2005**, *438*, 932.
- [68] M. M. Hickey, M. C. Simon, *Curr Top Dev Biol* **2006**, *76*, 217.
- [69] G. C. Gurtner, S. Werner, Y. Barrandon, M. T. Longaker, Nature 2008, 453, 314.
- [70] L. A. DiPietro, *J Leukoc Biol* **2016**, *100*, 979.
- [71] A. M. Isidori, M. A. Venneri, D. Fiore, J Endocrinol Invest **2016**, *39*, 1235.
- [72] H. Kämpfer, J. Pfeilschifter, S. Frank, *Expressional Regulation of Angiopoietin-1 and-2 and the Tie-1 and-2 Receptor Tyrosine Kinases during Cutaneous Wound Healing: A Comparative Study of Normal and Impaired Repair*, **2001**.
- [73] C. Hellberg, A. Ostman, C.-H. Heldin, *Recent Results Cancer Res* **2010**, *180*, 103.
- [74] H. D. Beer, M. T. Longaker, S. Werner, *J Invest Dermatol* **1997**, *109*, 132.
- [75] R. D. Galiano, O. M. Tepper, C. R. Pelo, K. A. Bhatt, M. Callaghan, N. Bastidas, S. Bunting, H. G. Steinmetz, G. C. Gurtner, Am J Pathol 2004, 164, 1935.
- [76] W. Qi, C. Yang, Z. Dai, D. Che, J. Feng, Y. Mao, R. Cheng, Z. Wang, X. He, T. Zhou, X. Gu, L. Yan,
  X. Yang, J.-X. Ma, G. Gao, *Diabetes* 2015, *64*, 1407.
- [77] N. Urao, U. A. Okonkwo, M. M. Fang, Z. W. Zhuang, T. J. Koh, L. A. DiPietro, *Microvasc Res* 2016, 106, 57.
- [78] S. Sangiorgi, A. Manelli, M. Reguzzoni, M. Ronga, M. Protasoni, C. Dell'Orbo, *Anat Rec* (*Hoboken*) **2010**, *293*, 1639.
- [79] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [80] E. L. Campbell, W. J. Bruyninckx, C. J. Kelly, L. E. Glover, E. N. McNamee, B. E. Bowers, A. J. Bayless, M. Scully, B. J. Saeedi, L. Golden-Mason, S. F. Ehrentraut, V. F. Curtis, A. Burgess, J. F. Garvey, A. Sorensen, R. Nemenoff, P. Jedlicka, C. T. Taylor, D. J. Kominsky, S. P. Colgan, *Immunity* **2014**, *40*, 66.
- [81] S. R. Walmsley, C. Print, N. Farahi, C. Peyssonnaux, R. S. Johnson, T. Cramer, A. Sobolewski, A. M. Condliffe, A. S. Cowburn, N. Johnson, E. R. Chilvers, *Journal of Experimental Medicine* 2005, *201*, 105.
- [82] R. J. Butterworth, M. D. Thesis, THE HISTOLOGY OF HUMAN GRANULATING WOUNDS, 1992.
- [83] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [84] C. G. Elliott, J. Wang, J. T. Walker, S. Michelsons, J. Dunmore-Buyze, M. Drangova, A. Leask, D.
  W. Hamilton, *Tissue Eng Part A* 2019, 25, 1326.
- [85] D. W. Siemann, *Cancer Treat Rev* **2011**, *37*, 63.
- MSc. Thesis April 2024

- [86] D. J. Kominsky, E. L. Campbell, S. P. Colgan, *The Journal of Immunology* **2010**, *184*, 4062.
- [87] F. L. Cousins, A. A. Murray, J. P. Scanlon, P. T. K. Saunders, BMC Res Notes 2016, 9, DOI 10.1186/s13104-016-1842-8.
- [88] S. Kang, D. Lee, B. E. Theusch, C. J. Arpey, T. J. Brennan, Wound Repair and Regeneration 2013, 21, 730.
- [89] B. Bedogni, M. B. Powell, *Pigment Cell Melanoma Res* **2009**, *22*, 166.
- [90] V. Flacher, F. Sparber, C. H. Tripp, N. Romani, P. Stoitzner, in *Cancer Immunology, Immunotherapy*, 2009, pp. 1137–1147.
- [91] A. Stachura, I. Khanna, P. Krysiak, W. Paskal, P. Włodarski, Int J Mol Sci 2022, 23, DOI 10.3390/ijms23158621.
- [92] A. Bavelloni, G. Ramazzotti, A. Poli, M. Piazzi, E. Focaccia, W. Blalock, I. Faenza, *Anticancer Res* **2017**, *37*, 6511.
- S. Narayanan, S. Eliasson Angelstig, C. Xu, J. Grünler, A. Zhao, W. Zhu, N. Xu Landén, M.
  Ståhle, J. Zhang, M. Ivan, R. G. Maltesen, I. R. Botusan, N. Rajamand Ekberg, X. Zheng, S. B.
  Catrina, *Commun Biol* 2020, *3*, DOI 10.1038/s42003-020-01495-y.
- [94] S.-B. Catrina, K. Okamoto, T. Pereira, K. Brismar, L. Poellinger, *Hyperglycemia Regulates Hypoxia-Inducible Factor-1 Protein Stability and Function*, **2004**.
- [95] U. A. Okonkwo, L. Chen, D. Ma, V. A. Haywood, M. Barakat, N. Urao, L. A. DiPietro, *PLoS One* 2020, 15, DOI 10.1371/journal.pone.0231962.
- [96] J. Todoric, M. Löffler, J. Huber, M. Bilban, M. Reimers, A. Kadl, M. Zeyda, W. Waldhäusl, T. M. Stulnig, *Diabetologia* 2006, 49, 2109.
- [97] S. W. Hartwell, SURGICAL WOUNDS IN HUMAN BEINGS A HISTOLOGIC STUDY OF HEALING WITH PRACTICAL APPLICATIONS : II. FIBROUS HEALING, n.d.
- [98] A. F. McGettrick, L. A. J. O'Neill, *Cell Metab* **2020**, *32*, 524.
- [99] A. C. D. O. Gonzalez, Z. D. A. Andrade, T. F. Costa, A. R. A. P. Medrado, An Bras Dermatol 2016, 91, 614.
- [100] T. Cramer, Y. Yamanishi, B. E. Clausen, I. Förster, R. Pawlinski, N. Mackman, V. H. Haase, R. Jaenisch, M. Corr, V. Nizet, G. S. Firestein, H.-P. Gerber, N. Ferrara, R. S. Johnson, *HIF-1α Is Essential for Myeloid Cell-Mediated Inflammation*, **n.d.**
- [101] A. P. Sawaya, R. C. Stone, S. R. Brooks, I. Pastar, I. Jozic, K. Hasneen, K. O'Neill, S. Mehdizadeh, C. R. Head, N. Strbo, M. I. Morasso, M. Tomic-Canic, *Nat Commun* **2020**, *11*, DOI 10.1038/s41467-020-18276-0.
- [102] L. A. DiPietro, *J Leukoc Biol* **2016**, *100*, 979.
- [103] J. Wang, D. Tinney, M. Grynyshyn, J. G. Pickering, A. Power, L. Dubois, D. W. Hamilton, Wound Repair and Regeneration 2023, 31, 321.
- [104] **N.d.**

- [105] W. X. Hong, M. S. Hu, M. Esquivel, G. Y. Liang, R. C. Rennert, A. McArdle, K. J. Paik, D. Duscher, G. C. Gurtner, H. P. Lorenz, M. T. Longaker, *Adv Wound Care (New Rochelle)* 2014, *3*, 390.
- [106] H. Thangarajah, D. Yao, E. I. Chang, Y. Shi, L. Jazayeri, I. N. Vial, R. D. Galiano, X.-L. Du, R. Grogan, M. G. Galvez, M. Januszyk, M. Brownlee, G. C. Gurtner, *The Molecular Basis for Impaired Hypoxia-Induced VEGF Expression in Diabetic Tissues*, n.d.
- [107] L. Gould, P. Abadir, H. Brem, M. Carter, T. Conner-Kerr, J. Davidson, L. Dipietro, V. Falanga, C. Fife, S. Gardner, E. Grice, J. Harmon, W. R. Hazzard, K. P. High, P. Houghton, N. Jacobson, R. S. Kirsner, E. J. Kovacs, D. Margolis, F. McFarland Horne, M. J. Reed, D. H. Sullivan, S. Thom, M. Tomic-Canic, J. Walston, J. A. Whitney, J. Williams, S. Zieman, K. Schmader, J Am Geriatr Soc 2015, 63, 427.
- [108] L. Pradhan, C. Nabzdyk, N. D. Andersen, F. W. LoGerfo, A. Veves, *Expert Rev Mol Med* 2009, 11, DOI 10.1017/S1462399409000945.
- [109] E. Groppa, A. Colliva, R. Vuerich, T. Kocijan, S. Zacchigna, Int J Mol Sci 2020, 21, 1.
- [110] L. Gould, P. Abadir, H. Brem, M. Carter, T. Conner-Kerr, J. Davidson, L. Dipietro, V. Falanga, C. Fife, S. Gardner, E. Grice, J. Harmon, W. R. Hazzard, K. P. High, P. Houghton, N. Jacobson, R. S. Kirsner, E. J. Kovacs, D. Margolis, F. McFarland Horne, M. J. Reed, D. H. Sullivan, S. Thom, M. Tomic-Canic, J. Walston, J. A. Whitney, J. Williams, S. Zieman, K. Schmader, J Am Geriatr Soc 2015, 63, 427.
- [111] J. Wang, Cell Tissue Res **2018**, 371, 531.
- [112] A. J. Varghese, S. Gulyas, J. K. Mohindra, *Cancer Res* **1976**, *36*, 3761.
- [113] S. Kang, D. Lee, B. E. Theusch, C. J. Arpey, T. J. Brennan, *Wound Repair and Regeneration* **2013**, *21*, 730.
- [114] F. L. Cousins, A. A. Murray, J. P. Scanlon, P. T. K. Saunders, BMC Res Notes 2016, 9, DOI 10.1186/s13104-016-1842-8.
- [115] S.-B. Catrina, K. Okamoto, T. Pereira, K. Brismar, L. Poellinger, *Hyperglycemia Regulates Hypoxia-Inducible Factor-1 Protein Stability and Function*, **2004**.
- [116] Y. Guan, H. Niu, Z. Liu, Y. Dang, J. Shen, M. Zayed, L. Ma, J. Guan, H E A L T H A N D M E D I C I N E Sustained Oxygenation Accelerates Diabetic Wound Healing by Promoting Epithelialization and Angiogenesis and Decreasing Inflammation, 2021.
- [117] R. M. Leach, D. F. Treacher, ABC of Oxygen Oxygen Transport-2. Tissue Hypoxia, n.d.
- [118] H. Brem, O. Stojadinovic, R. F. Diegelmann, H. Entero, B. Lee, I. Pastar, M. Golinko, H. Rosenberg, M. Tomic-Canic, *Molecular Medicine* 2007, 13, 30.
- [119] Y. Guan, H. Niu, Z. Liu, Y. Dang, J. Shen, M. Zayed, L. Ma, J. Guan, H E A L T H A N D M E D I C I N E Sustained Oxygenation Accelerates Diabetic Wound Healing by Promoting Epithelialization and Angiogenesis and Decreasing Inflammation, 2021.

- [120] S. Narayanan, S. Eliasson Angelstig, C. Xu, J. Grünler, A. Zhao, W. Zhu, N. Xu Landén, M. Ståhle, J. Zhang, M. Ivan, R. G. Maltesen, I. R. Botusan, N. Rajamand Ekberg, X. Zheng, S. B. Catrina, *Commun Biol* 2020, 3, DOI 10.1038/s42003-020-01495-y.
- [121] H. Thangarajah, D. Yao, E. I. Chang, Y. Shi, L. Jazayeri, I. N. Vial, R. D. Galiano, X.-L. Du, R. Grogan, M. G. Galvez, M. Januszyk, M. Brownlee, G. C. Gurtner, *The Molecular Basis for Impaired Hypoxia-Induced VEGF Expression in Diabetic Tissues*, n.d.
- [122] T. Cramer, Y. Yamanishi, B. E. Clausen, I. Förster, R. Pawlinski, N. Mackman, V. H. Haase, R. Jaenisch, M. Corr, V. Nizet, G. S. Firestein, H.-P. Gerber, N. Ferrara, R. S. Johnson, *HIF-1α Is Essential for Myeloid Cell-Mediated Inflammation*, **n.d.**
- [123] J. Song, X. Yang, L.-J. Yan, *Hypoxia* **2019**, *Volume* 7, 33.
- [124] S. Taneja, S. W. Vetter, E. Leclerc, Int J Mol Sci 2021, 22, DOI 10.3390/ijms22158153.
- [125] V. P. Singh, A. Bali, N. Singh, A. S. Jaggi, *Korean Journal of Physiology and Pharmacology* **2014**, *18*, 1.
- [126] G. Zaccagnini, S. Greco, C. Voellenkle, C. Gaetano, F. Martelli, *Antioxid Redox Signal* 2022, 36, 685.
- [127] A. Bavelloni, G. Ramazzotti, A. Poli, M. Piazzi, E. Focaccia, W. Blalock, I. Faenza, Anticancer Res 2017, 37, 6511.
- [128] S. Biswas, S. Roy, J. Banerjee, S. R. A. Hussain, S. Khanna, G. Meenakshisundaram, P. Kuppusamy, A. Friedman, C. K. Sen, *Proc Natl Acad Sci U S A* **2010**, *107*, 6976.
- [129] A. P. Sawaya, R. C. Stone, S. R. Brooks, I. Pastar, I. Jozic, K. Hasneen, K. O'Neill, S. Mehdizadeh,
  C. R. Head, N. Strbo, M. I. Morasso, M. Tomic-Canic, *Nat Commun* **2020**, *11*, DOI 10.1038/s41467-020-18276-0.
- [130] C. Wetzler, H. K. Empfer, B. Stallmeyer, J. Pfeilschifter, S. Frank, Large and Sustained Induction of Chemokines during Impaired Wound Healing in the Genetically Diabetic Mouse: Prolonged Persistence of Neutrophils and Macrophages during the Late Phase of Repair, n.d.
- [131] T. A. Wilgus, S. Roy, J. C. McDaniel, Adv Wound Care (New Rochelle) **2013**, *2*, 379.
- [132] P. K. Barman, T. J. Koh, Front Cell Dev Biol 2020, 8, DOI 10.3389/fcell.2020.00528.
- [133] F. O. Quenum Zangbede, A. Chauhan, J. Sharma, B. B. Mishra, *Journal of Neuroscience* 2018, 38, 6737.
- [134] U. A. Okonkwo, L. Chen, D. Ma, V. A. Haywood, M. Barakat, N. Urao, L. A. DiPietro, *PLoS One* 2020, 15, DOI 10.1371/journal.pone.0231962.
- [135] H. Kämpfer, J. Pfeilschifter, S. Frank, Expressional Regulation of Angiopoietin-1 and-2 and the Tie-1 and-2 Receptor Tyrosine Kinases during Cutaneous Wound Healing: A Comparative Study of Normal and Impaired Repair, 2001.
- [136] N. Jetten, S. Verbruggen, M. J. Gijbels, M. J. Post, M. P. J. De Winther, M. M. P. C. Donners, Angiogenesis 2014, 17, 109.

- [137] S. Wang, G. Liu, Y. Li, Y. Pan, Front Immunol **2022**, 13, DOI 10.3389/fimmu.2022.840029.
- [138] S. Elliot, T. C. Wikramanayake, I. Jozic, M. Tomic-Canic, *Journal of Investigative Dermatology* **2018**, *138*, 736.
- [139] P. Huynh, J. Phie, S. M. Krishna, J. Golledge, *BMJ Open Diabetes Res Care* 2020, 8, DOI 10.1136/bmjdrc-2019-000982.
- [140] R. C. Fang, Z. B. Kryger, D. W. Buck, M. De La Garza, R. D. Galiano, T. A. Mustoe, *Wound Repair and Regeneration* **2010**, *18*, 605.
- [141] L. Dunn, H. C. G. Prosser, J. T. M. Tan, L. Z. Vanags, M. K. C. Ng, C. A. Bursill, Journal of Visualized Experiments 2013, 2013, 1.
- [142] S. J. Jackson, N. Andrews, D. Ball, I. Bellantuono, J. Gray, L. Hachoumi, A. Holmes, J. Latcham, A. Petrie, P. Potter, A. Rice, A. Ritchie, M. Stewart, C. Strepka, M. Yeoman, K. Chapman, *Lab Anim* **2017**, *51*, 160.

# **Chapter 3**

# **General Discussion**

## **3.1 Summary**

In this thesis, we hypothesized that neutrophil persistence would result in significantly increased oxygen consumption in the wound bed of the db/db mouse, compared to control. This increased stress would manifest as delayed resolution of hypoxia, delayed wound closure, enhanced neutrophil persistence, and reduced formation of diabetic neo-vasculature.

Next, we present a summary of our findings in the context of our aims.

<u>Aim 1:</u> Investigate the relationship between persistence of neutrophils and delayed hypoxia resolution in the context of impaired diabetic healing.

**Objective:** Confirm that delayed hypoxia resolution correlates with increased persistence of neutrophils in the diabetic wound bed.

We were not able to establish a relationship between delayed hypoxia resolution and increased persistence of neutrophils in the wound bed in the db/db mouse. Although we observed a significantly increased number of neutrophils in the db/db wounds beds at day 7 (Figure 8c), this was not associated with increased hypoxic load (Figure 7c). In fact, hypoxia was significantly increased in the wild-type at both timepoints assessed day 3 and 7. At day 3, there were more hypoxic cells in the wound beds and one of the edges of wild type mice and at day 7 we observed a significantly elevated hypoxic load in the wild type wound beds only (Figure 7c).

<u>Aim 2:</u> Investigate the relationship between hypoxia resolution and the rate of revascularization of tissues.

**Objective:** Confirm that as the wound re-establishes its vascular network, the tissue becomes less hypoxic.

We were not able to demonstrate a temporal decrease in the gradient of hypoxia as the tissue became re-vascularized. Although we observed a significant increase in the number of CD31<sup>+</sup> cells in the wild type at day 3 (Figure 9c), this was not accompanied by an overall decrease in hypoxic load (Figure 7c). There were no significant differences in re-vascularization at day 7 (Figure 9c), and wild-types again presented with a higher hypoxic load at day 7 in the wound beds.

<u>Aim 3:</u> Validate the mouse model as one re-capitulating the major defects seen in impaired human wound healing.

**Objective**: Confirm that the mouse exhibits appropriately comparable wound healing defects to the human.

We were able to validate the db/db model as one of impaired wound healing across a number of known skin healing impairments in the human. Db/db mice presented with a significant delay in closure kinetics at days 3, 7 and 12 (Figure 1), exhibited delayed re-epithelialization at day 7 (Figure 2), and exhibited reduced collagen deposition at day 7 (Figure 4). Additionally, db/db mice presented with a significantly elevated neutrophil load at day 7 (Figure 8c), a finding often reported in impaired human healing literature <sup>[108]</sup>. Lastly, db/db mice exhibited significantly less re-vascularization at day 3 (Figure 9c), a phenomenon well characterized in impaired human healing literature <sup>[109]</sup>.

# **3.2 Contributions to the Literature**

#### **3.2.1 Oxygen and the Healing Wound**

Oxygen is known to be a pre-requisite for proper wound healing <sup>[83]</sup>. Reparative processes such as decontamination/debridement, cell proliferation, collagen synthesis and angiogenesis all impart a significant increase in metabolic demand on the healing wound tissue <sup>[83]</sup>. Oxygen is critical for the production of adenosine triphosphate (ATP), which is the biological equivalent of energy <sup>[83]</sup>. Therefore, sufficient tissue oxygenation is a necessary to meet increased energy demands, facilitating proper cellular function. Due to full-thickness dermal injury, blood vessels are damaged or lost, leading to a loss of tissue perfusion accompanied by a sharp increase in local hypoxia <sup>[83]</sup>. This rise in early acute hypoxia is responsible for initiating the earliest steps in dermal tissue repair <sup>[83]</sup>. It does so by boosting reactive oxygen species (ROS) activity, by recruiting platelets and endothelial cells and by stimulating cytokine release from platelets, monocytes and parenchymal cells <sup>[83]</sup>. ROS stimulate cytokine and chemokine release, the primary effects of which are the recruitment of neutrophils and macrophages and activation of fibroblasts <sup>[83]</sup>.

Although what constitutes the exact length of acute hypoxia is not completely characterized, it is generally recognized that acute hypoxia is a necessary initiator of wound repair, while prolonged, chronic hypoxia has a detrimental effect on nearly all aspects of normal wound repair <sup>[83]</sup>. In our present study, we focused on quantifying any potential differences in the temporal profile of hypoxic cell load in full-thickness murine wounds between the db/db and wild-type murine phenotype. We hypothesized that the hypoxic load would decrease as the wound progressed to the angiogenic phase, and that correlations would be evident between neo-angiogenesis and neutrophil persistence. The secondary aim of the project was validation

of the db/db murine model in the context of diabetic skin healing, and how closely it recapitulated what is commonly described in the diabetic patient. Evaluation of the db/db model from the standpoint of hypoxic healing, neutrophil persistence and decreased angiogenesis is an important factor given the commonly reported wound healing deficits reported in diabetic humans in the literature<sup>[110]</sup>. Of significance, we also focused on whether murine models of skin healing replicate the data derived by Rosalind Butterworth, namely the finding of neutrophil persistence in wounds up to 12 weeks old even after closure <sup>[82]</sup>. After taking serial punch biopsies from three groups of pilonidal sinus wounds, she reported the neutrophil numbers to be similar irrespective of the timepoint evaluated. Although we recognize this data to stand in contrast with the general consensus in the field - that beyond the acute inflammatory phase, neutrophils serve little beneficial functions and their levels generally fall after about day 5 of healing <sup>[111]</sup>, we aimed to interrogate our murine model with the goal of observing whether it would replicate the data reported by Butterworth.

#### 3.2.2 Tools for Quantifying Cellular Hypoxia: Hypoxyprobe<sup>TM</sup>

In 1976, Varghese et al. demonstrated that 2-nitroimidazoles form adducts in hypoxic cells *in vitro* and *in vivo*. They subsequently showed that these adducts are formed with thiol groups in proteins, peptides and amino acids in such a way, that the entire structure of 2-nitroimidazole is retained, thus opening the door on the possibility of labeling and detecting these adducts <sup>[112]</sup>. It was of significance demonstrated that an oxygen tension of  $pO_2 < 10$  mmHg *was required* for adduct formation, meaning cells had to be in hypoxic conditions. This binding was dependent on the presence of redox enzymes in cells, however, pimonidazole binding was not dependent on the presence of any specific redox enzyme. In

animal. It is a substituted 2-nitroimidazole whose chemical name and sole component is pimonidazole hydrochloride. Hypoxyprobe<sup>TM</sup> relies on pimonidazole hydrochloride as its hypoxia marker. It forms protein adducts only in cells perceiving a pO<sub>2</sub> < 10 mmHg. It is these adducts, that are subsequently visualized using a specific monoclonal antibody. Pimonidazole detection has been demonstrated to be a reliable and accurate measure of tissue hypoxia <sup>[113,114]</sup>. Importantly, the probe has also been shown to be resistant to fixation and can still be detected <sup>[114]</sup>.

Validation of Hypoxyprobe<sup>TM</sup> has been documented in models of endometrial shedding, where Cousins et al. demonstrated intense immunostaining for hypoxia during the breakdown and shedding of the decidual mass within 24 hours of progesterone withdrawal <sup>[114]</sup>. This group was able to show a gradient of hypoxia developing temporally as the endometrial vessels regressed and the lining was shed. Validation of this methodology in the context of tissue repair comes from the work by Kang et al <sup>[113]</sup>. This group harvested tissue from rats for pimonidazole processing, following a gastrocnemius incision or hindpaw incision at days 1, 2, 4 and 10, using the opposite side as a control. They reported temporal changes in the amount of hypoxic labeling, with staining being most intense at days 1 and 2, diffuse staining at day 4 and little staining by day 10. Of significance, both models showed that intensity of staining and gradients can be detected using this method.

In our study, we utilized Hypoxyprobe<sup>TM</sup> to label and quantify populations of hypoxic cells in a murine full-thickness excisional wound. To our knowledge, this is the first published report of the use of Hypoxyprobe<sup>TM</sup> in assessing hypoxic cells in full-thickness excisional

murine skin healing. Hypoxyprobe<sup>TM</sup> was administered intraperitoneally, in a saline solution and animals were sacrificed 90 minutes later, to allow the probe to circulate and bind to targets. Positive controls were obtained in the form of murine kidney (Appendix 4.7).

### 3.2.3 Hypoxia and Tissue Response During Skin Healing

Our work demonstrated that wild-type mice presented with a significantly elevated number of hypoxic cells at day 3 on the right wound edge as well as in the wound bed itself (Figure 7c), and in the day 7 wound bed alone (Figure 7c), when compared to db/db mice. The differences were greater at day 3 (p=0.0179/p=0.0048) versus day 7 (p=0.0238), consistent with previous work in the field describing the impaired adaptation to hypoxia in hyperglycemic wounds <sup>[115]</sup>. Although it is commonly reported that both human and diabetic mouse wounds are more hypoxic when compared to healthy tissues or wild-type animals respectively <sup>[83]</sup>, if a defect exists in the mechanism of cellular hypoxic sensing/adaptation, wounded tissue in diabetics may be hypoxic, but this may not be sensed by the resident cells, thus accounting for the lower hypoxic signal seen in this population. At day 7, we observed a significantly greater percentage of hypoxic cells in the wound beds of wild-type animals however this significant difference was small compared to day 3. We postulate that by day 7, the db/db mice may begin to recover, similar to the wild-types with regard to hypoxic sensing and adaptation. However, due to the delay observed at day 3, their entire wound healing process may already be delayed due to insufficient initial recruitment of reparative cell populations as a result of insufficient early hypoxia recognition and subsequent adaptation. Interestingly, the increased hypoxic load at day 3 and 7 in wild-types, correlated with improved healing outcomes including significantly faster wound closure kinetics at days 3, 7

and 12 (Figure 1), increased re-epithelialization at day 7, as evaluated by Mason's Trichrome stain (Figure 2) and increased wound bed collagen deposition at day 7, as evaluated by Picrosirius Red stain (Figure 4).

We next assessed specifically the degree of epithelial hypoxia via Hypoxyprobe<sup>™</sup>, by tracing the length of the signal from the edges of the wound, until the first obvious drop in signal intensity. Generally, the epithelium stained strongly at the immediate wound edges of both phenotypes, and the signal gradually diminished laterally away from the wound towards uninjured tissue. Isolated pockets of hypoxic signaling were evident in epithelial areas far from the wound, but these were generally associated with the presence of closely underlying hair follicles, which are known to be hypoxic <sup>[89]</sup>. We did not see significant differences at either day 3 (Figure 6c) or 7 (Figure 6c) between phenotypes, suggesting that the proposed defect in hypoxia adaptation may not play as prominent of a role in epithelial cells, as both phenotypes exhibited a closely comparable pattern of staining. Our hypoxia related findings with respect to epithelium were inconsistent with previous work in the field, namely the postulation that diabetic wounds are generally more hypoxic <sup>[116]</sup>. Although this may be true on a tissue perfusion level, this may not necessarily translate to the resident surface cells of the tissue <sup>[117]</sup>. It must be noted that we don't suggest that either left or right sides show significant differences in cellular infiltration. These regions were assigned the left or right label based solely off of the slide image on the screen for quantification purposes. Work by Brem et al., demonstrated that distinct wound regions have specific histological profiles <sup>[118]</sup>. As such although we considered combining the data from the edges, we decided to keep it separate so as not to lose any potential differences existing in different wound locations. We do not postulate that the left or right side accounts for the difference, but rather we demonstrate a distinct wound region that exhibited a significant difference.

MSc. Thesis – April 2024

Our work demonstrates for the first time, an increased number of hypoxic cells in wild-type full-thickness excisional wounds at both timepoints compared to the wounds created in db/db mice. This finding, although counter to what is reported in the field <sup>[119,120]</sup> is reconciled in the context of well characterized defective diabetic hypoxia adaptation/sensing <sup>[115,121,122]</sup>, a failure of early recruitment of reparative cells, and the delayed healing outcomes we observed in the db/db mouse.

Common causative factors of local wound tissue hypoxia in the diabetic patient include alterations of the vascular bed, atherosclerosis, hypertension, and swelling <sup>[83]</sup>. Hypoxia plays an important role in many diabetic complications. Hyperglycemia has also been postulated to induce a pseudohypoxia state <sup>[123]</sup>. This postulation is based on data showing that high glucose concentrations induce a high NADH<sup>+</sup>/NAD<sup>+</sup> ratio in cells under normoxic conditions<sup>[115]</sup>. In addition to this effect, studies have shown that hyperglycemia has a strong potential to interfere with cellular responses and adaptation to hypoxia. Adaptive cellular responses to hypoxia are mediated by HIF-1, a heterodimeric transcription factor made up of two subunits<sup>[115]</sup>. Modulation of HIF-1 activity is critically dependent on the breakdown of the HIF-1 $\alpha$  subunit under normoxic conditions <sup>[115]</sup>. "The molecular basis of its breakdown is the O<sub>2</sub> dependent hydroxylation of one of the two proline residues, in the oxygen dependent degradation domain of HIF-1 $\alpha$  by a specific prolyl 4-hydroxylase"<sup>[115]</sup>. In this state, HIF-1 $\alpha$ binds to the von Hippel-Lindau tumor suppressor protein, which acts as an E3 ubiquitin ligase and targets HIF-1 $\alpha$  for proteasomal degradation <sup>[115]</sup>. Under hypoxic conditions, HIF- $1\alpha$  is stabilized against degradation and upregulates a series of genes involved in

angiogenesis, glycolytic energy metabolism, cell proliferation and survival"<sup>[115]</sup>. Although hypoxia is the main regulator of HIF-1 $\alpha$ , other factors involved in diabetic pathogenesis have the potential to indirectly influence cellular hypoxic adaptation<sup>[124]</sup>. One of these factors is advanced glycation end products (AGEs), involved in many diabetic complications <sup>[125]</sup>. Work by Catrina et al. 2004, demonstrated that aberrant glucose levels can inhibit hypoxiainduced upregulation of HIF-1 $\alpha$  protein levels in human dermal fibroblasts and human dermal microvascular endothelial cells <sup>[115]</sup>. This effect was also reported to be dose dependent. These cells also showed reduced levels of HIF-1 $\alpha$  protein stabilization when compared to those grown in normal glucose concentrations. They were also able to validate their results in vivo, by showing that diabetic foot ulcers (DFU) expressed lower levels of HIF-1 $\alpha$ , compared with chronic venous ulcers<sup>[115]</sup>. Overall, this group demonstrated that hyperglycemia interferes with protection from proteasome-mediated degradation of HIF-1 $\alpha$ . If HIF-1 $\alpha$  is prematurely degraded in conditions of hyperglycemia, this could plausibly interfere with diabetic cellular ability to sense and adapt to hypoxic stimuli. It was interesting that the *in vivo* effects were observed in the DFU, but not in chronic venous ulcers, further strengthening the likelihood of hyperglycemia being the responsible factor. This data suggests an interaction mechanism between two of the most important determinants of the chronic complications commonly seen in diabetes – hyperglycemia and hypoxia [115,119]

A role for microRNAs (miRNAs) in the context of diabetic healing has also been documented with respect to hypoxia<sup>[120,126]</sup>. miRNAs are a group of tightly conserved noncoding RNAs that are known to have the ability to either enhance mRNA degradation or inhibit post-transcriptional translation <sup>[127]</sup>. Dysregulated expression of a single miRNA can drastically alter the translation of multiple genes in a cell, with the potential to alter the cell phenotype <sup>[127]</sup>. In particular, miRNA-210 has been extensively investigated and identified *MSc. Thesis – April 2024* 96
as a major miRNA induced under hypoxic conditions. Several studies claim a direct interaction between miR-210 expression and hypoxia <sup>[127][120,126,128]</sup>, and moreover several investigative groups report miR-210 as one of the highest upregulated miRNAs in hypoxic cells. miR-210 is also known to be involved in the regulation of expression of a cluster of genes involved in processes highly important for energy metabolism and tissue repair <sup>[120]</sup>. These processes include angiogenesis, cell proliferation and metabolic adaptation <sup>[120]</sup>. Work by Narayanan et al., demonstrated that induction of miR-210 is suppressed in the wounds of diabetic patients and in the wounds of db/db mice <sup>[120]</sup>. In summary, these findings support the theory of an insufficient adaptive response to the presence of hypoxia in diabetic wounds either due to insufficient HIF-1 $\alpha$  protein stabilization, aberrant miR-210 expression or potentially undiscovered factors relating to hyperglycemia and the presence of AGEs. Narayanan et al., partially attributed a range of poor wound healing outcomes such as diminished proliferation, angiogenesis, migration and increased inflammation to miR-210 insufficiency, as a localized treatment with an miR-210 clone reduced inflammation, improved granulation, proliferation and angiogenesis in db/db wounds<sup>[120]</sup>.

In conclusion, the current evidence in the field with regard to hypoxic diabetic healing seems to strongly suggest a defect in cellular hypoxic adaptation, along with a variety of resulting undesirable outcomes in skin healing<sup>[120–122]</sup>. It is interesting that diabetic wounds are generally postulated to be more hypoxic <sup>[119]</sup>, yet if a defect exists in how the reparative cells sense and respond to hypoxia, it is entirely plausible to suggest a cyclical mechanism in which expression of HIF-1 $\alpha$  and miR-210 is dysregulated in the context of AGE accumulation, leading to an overall delay in healing, as one of the most important early mediators in healing is not activating in a timely manner, thus temporally delaying the entire wound healing process.

MSc. Thesis – April 2024

#### 3.2.4 Hypoxia and Inflammation

Our evidence showed no significant differences in neutrophil infiltrate at day 3 (Figure 8c) and a significantly elevated number of neutrophils at day 7 in the wound beds of db/db animals (Figure 8c). Although not statistically significant, wild-type animals had slightly higher counts in their wound beds and edges. This finding is consistent with previous work in the field describing a general increased inflammatory cell load in the diabetic healing context <sup>[101]</sup>. However, we postulate that this observation is a delay in early neutrophil recruitment, resulting in the need for an increased neutrophil count at day 7 in the wound bed in db/db mice to facilitate wound debridement. It is known that approximately 95% of myeloid cells are recruited to sites of inflammation, as opposed to residing there <sup>[122]</sup>. Next, we discuss a range of evidence that could potentially account for delayed neutrophil infiltration, or alternatively a fault in neutrophil clearance by macrophages.

Work by Sawaya et al., focused on the transcription factor FOXM1 specifically in the context of decreased inflammatory cell recruitment <sup>[129]</sup>. This transcription factor was identified as an important mediator of immune cell activation, recruitment and survival in normal healing <sup>[129]</sup>. This group first outlines an inflammatory transcriptional signature present in acute, healthy oral and skin wounds, and further goes to show that this same signature is deficient in DFUs <sup>[129]</sup>. More specifically, they found that levels of FOXM1, responsible for activation and recruitment of inflammatory cells was downregulated in diabetic patients. They confirmed these findings with evidence obtained using the db/db model, which showed downregulated levels of FOXM1 concomitant with delayed wound healing and a decrease in neutrophil and macrophage recruitment to the wound site <sup>[129]</sup>.

Our data is also supported by the findings of Wetzler et al., which showed that wound repair in the db/db mouse was characterized by a prolonged inflammatory response and a prolonged expression of macrophage inflammatory protein-2 and macrophage attractant protein-1 <sup>[130]</sup>. It was demonstrated that a sustained expression of the inflammatory cytokines IL-1B and TNF- $\alpha$  were present in the db/db mouse as late as day 13 after injury compared to the wildtypes which exhibited diminishing levels from day 5 onwards <sup>[130]</sup>. As neutrophils and macrophages are both potent producers of these cytokines, this group hypothesized that healing in the db/db mouse might be accompanied by increased numbers of neutrophils and macrophages during the late phase of repair. Their subsequent data confirmed this hypothesis, with clearly elevated levels of both neutrophils and macrophages at the late stages of repair <sup>[130]</sup>. This data further supports our postulation that the inflammatory response in the db/db skin healing model is not due to a greater persistence of neutrophils in the healing wound <sup>[131]</sup>, but rather delayed inflammatory response.

Our data is also consistent with previously reported evidence in the field relating to insufficient hypoxia adaptation <sup>[122]</sup>. Hypoxia has been demonstrated to have a profound effect on a broad range of myeloid cell properties, including migration, cytokine secretion and survival <sup>[122]</sup>. It has long been described that neutrophils are highly dependent on anaerobic glycolysis for ATP production, and are thus highly adapted to functioning in a hypoxic environment <sup>[122]</sup>. It has also been shown that inhibition of glycolysis leads to direct inhibition of ATP production and consequently negatively impacts such inflammatory cell properties as adhesion, extravasation, motility and invasion <sup>[122]</sup>. Hypoxia is also a key regulator of glycolysis via the transcription factor HIF-1 $\alpha$ , thus we hypothesize that the previously discussed insufficiency in HIF-1 $\alpha$  described in murine and human diabetic tissues can lead to insufficient ATP levels and an overall deficiency in early neutrophil recruitment,

thus delaying the entire skin healing process in the db/db animal <sup>[122]</sup>. Interestingly, work by Cramer et al. demonstrated that the loss of HIF-1 $\alpha$  in the myeloid lineage results in an almost complete lack of the inflammatory response in the skin <sup>[122]</sup>. They described a pronounced inflammatory response in wild-type animals, with prominent inflammatory cell infiltration and epidermal hyperproliferation, which were absent in the HIF-1 $\alpha$  knockouts. This work lends further credibility to the postulation of insufficient hypoxic adaptation having potential downstream effects on neutrophil recruitment and metabolism, leading to a delay in infiltration with significantly higher numbers at day 7 granulation tissue of db/db animals (Figure 8c).

While transcriptions factors activate specific genes in wound healing, work by Barman & Koh., proposes an alternative mechanism accounting for delayed neutrophil clearance. Their work expands on the concept of macrophage polarization, with an imbalance in proinflammatory M1 macrophages and pro-reparative M2 macrophages a possible cause for insufficient neutrophil clearance and delayed resolution of inflammation <sup>[132]</sup>. Although the work attempting to uncover the microenvironmental stimuli that cause altered macrophage polarization is largely unknown in vivo, it is known that this shift has to occur for physiological healing to proceed normally <sup>[132]</sup>. It is also known that the process of efferocytosis, or neutrophil clearance is primarily carried out by M2 macrophages <sup>[133]</sup>. Thus, if the polarization towards an M2 phenotype does not occur in db/db skin wounds, this could account for a decreased rate of neutrophil clearance from the granulation tissue. This provides an alternative explanation for the significantly increased neutrophil load at day 7 in the db/db animal we observed, as being not a result of delayed infiltration but a delayed clearance of these cells from the wound bed. Further work is needed to attempt to elucidate

which mechanism may be primarily responsible for the overall sustain in inflammation seen in the diabetic, or whether it is a combination of both.

In summary our findings of significantly increased neutrophil numbers in the wound beds of db/db animals at day 7 are in agreement with the current consensus of potential factors accounting for a delayed inflammatory phase resolution seen in the db/db animal. The potential mechanisms include - insufficient activity in myeloid cell activating transcriptional factors such as FOXM1 or HIF-1 $\alpha$ , which are essential in the earliest stages of inflammatory cell recruitment for neutrophils to generate sufficient energy to extravasate, migrate and carry out their functions or a potential defect in the clearance of these cells once their functions are completed.

#### 3.2.5 Hypoxia and Angiogenesis

We observed a significant decrease in the number of CD31<sup>+</sup> cells at day 3 in the db/db mouse at the right edge and wound bed (Figure 9c). Interestingly, this observation correlated with a significantly higher number of hypoxic cells in the same regions in the wild type, suggesting a correlation between hypoxia and endothelial cell recruitment. By day 7, the number of CD31<sup>+</sup> cells was similar between phenotypes (Figure 9c), suggesting that either the db/db mouse is able to partially overcome any existing angiogenic defects by day 7, or suggesting the possibility of involvement of other factors allowing it to compensate for defective hypoxia adaptation in the early stages of wound healing. This postulation is lent credibility by the fact that the variability in results between phenotypes at day 7 was quite low, with overall similar distributions of CD31<sup>+</sup> cells at both the edges and the wound bed. It is interesting that the higher level of hypoxia at day 7 in the wild type did not correlate with significantly elevated CD31<sup>+</sup> cell populations, further suggesting that the role of hypoxia

adaptation may be most prominent in the earliest, acute stages of healing after which other factors may become more important in regulating the wound healing events. Our data stands in partial contrast with what is commonly reported in the field, namely lower levels of VEGF with subsequent lower CD31<sup>+</sup> cell populations and impaired angiogenesis in the context of diabetic healing at timepoints at timepoints of day 10 and 18 and day 10 respectively <sup>[134]</sup>-

We suggest that the db/db model should be considered as a model of delayed rather than impaired healing, as most physiological processes occurring in the wild type, also occur in the diabetic at comparable levels, however, they were all delayed temporally. This model is commonly presented as a model of impaired healing, with researchers commonly citing defects in closure kinetics, angiogenic insufficiency and aberrant inflammation as proof of validation <sup>[130,134]</sup>. Our study was able to recapitulate most of these effects. The db/db mice exhibited delayed closure kinetics at days 3 7 and 12, they exhibited an angiogenic deficit at day 3, and exhibited significantly increased neutrophil numbers at day 7. The only parameter that this model did not recapitulate was the increased hypoxic load, with wild-type mice exhibiting more hypoxia at both days 3 and 7. This could raise potential issues with any hypoxia targeted treatments in the db/db model as we postulate that it exhibits a defect in hypoxic sensing/adaptation. If defects in hypoxia sensing exist in this model, it may not be an optimal model for testing of hypoxia targeted treatments, as the cells in the tissue may not perceive themselves as hypoxic. If hypoxia targeted treatments are to be tested in these studies, it can logically create problems with translation if the cells in the db/db wound are not adapting/responding to the environment that is to be addressed in the study.

Hypoxia activates the transcription factor HIF-1 $\alpha$ , which then binds to the HRE in the gene promoter region of vascular endothelial growth factor (VEGF), which in turn upregulates VEGF<sup>[83]</sup>. This factor is a major angiogenic factor that stimulates endothelial cells to migrate, proliferate and form new capillaries <sup>[83]</sup>. If hypoxia is one of the central initiators of angiogenesis in the healing wound, and diabetic cells fail to perceive or adapt to it due to defective HIF-1 $\alpha$  activity, this can plausibly result in a delay or failure in recruitment of adequate numbers of angiogenic cells. Work by Thangarajah et al. examined whether hypoxic upregulation of VEGF was impaired in diabetes. First, they demonstrated that normal fibroblasts isolated from healthy patients upregulated VEGF protein levels 2.2-fold in response to hypoxia, while fibroblasts from diabetic patients demonstrated no significant increase in VEGF production <sup>[121]</sup>. To confirm their findings were due to hyperglycemia, they then grew normal fibroblasts from patients with no vascular disease in low and high glucose conditions for 4 weeks, in order to simulate hyperglycemia. These cells were then placed in hypoxic conditions for 24 hours, and VEGF protein levels were quantified. They found that fibroblasts grown in low glucose upregulated VEGF protein levels by about 3.2-fold, while cells grown in high glucose showed a VEGF expression increase of only 1.2-fold. They paralleled these findings in murine cells by demonstrating a similar pattern of VEGF production in dermal fibroblasts grown in hyperglycemic conditions (4 weeks). When they cultured the same murine cells in an acute high glucose environment (<24 hours), they saw no disparities in VEGF upregulation relative to cells cultured in a low glucose environment, suggesting that chronic, and not acute exposure to hyperglycemia is the responsible factor in compromising VEGF expression <sup>[121]</sup>. They also demonstrated that db/db mice had significantly lower levels of VEGF expression in comparison to wild types at both 24 and 72 hours after injury. Lastly, this group demonstrated that chronic high glucose exposure

MSc. Thesis – April 2024

decreases HIF-1 $\alpha$  transactivation in response to hypoxia, thus plausibly implicating dysregulation of this transcription factor in all three main areas of interest in our project – hypoxia sensing, inflammation and angiogenesis. Interestingly, this group failed to confirm the findings of impaired HIF-1 $\alpha$  stability in a hyperglycemic setting that has been reported by other groups <sup>[115,122,128]</sup>. They went on to suggest that the hyperglycemia associated defect in HIF-mediated gene transcription results from a defect in HIF transactivation and not as a consequence of impaired HIF-1 $\alpha$  stability <sup>[121]</sup>. They supported this postulation by demonstrating that the interaction between HIF-1 $\alpha$  and its coactivator p300 is greatly decreased in hyperglycemic conditions. Thus, it may be pertinent to look not only at HIF-1 $\alpha$  activity and stability, but also its known co-activators as well.

Other research has highlighted other potential deficits responsible for angiogenic dysregulation commonly reported in the diabetic condition <sup>[103]</sup>. Recent work in our lab showed that it is not specifically reduced angiogenesis, but a premature terminal stabilization of the vasculature by pericytes and mural cells, thus preventing these vessels from undergoing further necessary maturation and remodeling processes <sup>[103]</sup>. The idea of a vascular "plasticity window" is not novel. It has been described by Benjamin et al., as a time during which neo-vasculature matures and undergoes remodeling, and this process was shown to be necessary for successful wound healing progression <sup>[104]</sup> As an alternative explanation to the HIF-1 $\alpha$  mediated insufficient recruitment of endothelial cells and induction of VEGF, we propose that it is not necessarily the number of new blood vessels in the wound that accounts for successful wound healing, but rather the amount of these vessels that undergo remodeling and maturation<sup>[103]</sup>. Additionally, building on the previously discussed imbalance between pro-inflammatory M1 macrophages and pro-reparative M2 macrophages present during diabetic healing, work by Jetten et al., demonstrated that it is the MSc. Thesis – April 2024 104

M2 macrophages which are chiefly responsible for promoting angiogenesis in a mouse model <sup>[136]</sup>. It is noteworthy that while M1 macrophages rely on glycolysis for energy, M2 macrophages require oxidative energy pathways <sup>[137]</sup>. If the diabetic wound is not adapting to hypoxia properly, and failing to initiate processes necessary to resolve it, this could provide an alternative explanation for decreased M2 macrophage populations seen in the diabetic condition, with the accompanying angiogenic defects as these cells are known to be major sources of TGF-B and VEGF both critically important for angiogenesis <sup>[136]</sup>.

Although it is early to suggest that dysregulation of any transcriptional factor, cytokine or cell population is solely responsible for the defects observed in the db/db mouse, namely delayed closure kinetics, insufficient angiogenesesis, reduced collagen deposition and reduced re-epithelialization in the present study, it is noteworthy that the great majority of our data could be explained by an insufficient early adaptation to hypoxic stimuli. As the aim of this project was the broad study of the relationship between hypoxia resolution, inflammation and angiogenesis, the intricate study of how cells become adapted/activated by hypoxia fell beyond its immediate scope. However, we suggest further work to focus closely on what transcription factors are temporally upregulated in hypoxic conditions and in what reparative cell populations. Future work should also aim to elucidate the temporal profile of the vascular plasticity window, as such findings could lead to better temporally targeted therapeutic interventions.

#### **3.2.6 Murine Model and it's Validation**

The main aspect to be considered when selecting any particular model is the degree of clinical relevancy or translation. The model used in the present study is the db/db strain, created by a point mutation in the leptin receptor gene <sup>[138]</sup>. Although a great body of work has been carried out with this and other models, murine models as a whole have shown limited translational value to impaired human wound healing conditions <sup>[138]</sup>. Furthermore, there is a recognized lack of a current consensus on the most suitable mouse model to be used in attempting to model diabetes associated ulceration <sup>[139]</sup>. The db/db model has reliably been established as one exhibiting impaired healing in an excisional wounding model and has thus achieved widespread acceptance in wound healing studies<sup>[140]</sup>. However, there are several important limitations of this model which will be discussed next. One of the most apparent limitations of this model is the greater role of leptin in murine appetite control when compared to humans <sup>[140]</sup>. This raises the concern that the deficiency of the leptin axis in mice may potentially have a wider range of phenotypic changes than what is clinically observed in human Type 2 diabetes. Experiments in leptin-deficient ob/ob mice have shown that leptin may function is an important modulator of the wound healing cascade, including mitogenic activity and angiogenesis <sup>[140]</sup>. Another major limitation of this model is the etiology of its obesity and subsequent diabetes development, as analogous forms of this severe type of condition in humans are rare <sup>[140]</sup>. Additionally, in humans Type 2 diabetes is a polygenic disease, involving many factors, while in mice this condition is modelled monogenically. Interestingly, although this model is popular due to its clear and reproducible wound healing defects, such as delayed contraction and re-epithelialization, this delay in wound contraction may in part be attributable to the striking obesity of this model <sup>[140]</sup>. This obesity maintains a constant stretch on the skin, plausibly leading to a reduction of available

loose skin required for optimal contraction to occur. As such, this particular wound healing deficit may be independent of the diabetic condition of the mouse, and is thus different in etiology to the delay observed in humans. Lastly, a recent report demonstrated that the impaired wound healing parameters in db/db mice may not be related to the level of hyperglycemia, thus suggesting that the impaired healing seen in the db/db mouse may not be readily translatable to the wound healing defects seen in patients with Type 2 diabetes <sup>[140]</sup>. As such, although the db/db model may provide value as a model of impaired contraction and delayed excisional wound closure, the extent to which it reflects similar defects in human Type 2 diabetes should remain open to question.

Ulcer development or wound chronicity in diabetics can take decades and is promoted by factors such as years of macro-vascular insufficiency, chronic inflammation and chronic hyperglycemia<sup>[121]</sup>. Understandably, the mouse lifespan may be too short to allow for accurate recapitulation of these prolonged processes. It is also noteworthy that rodent healing differs significantly from human healing. Mice heal primarily by contraction, whereas humans heal mainly by re-epithelialization <sup>[141]</sup>. Although researchers often attempt to overcome this by splinting the wound, previous work in our lab showed that the murine model may demonstrate a primarily contractile defect as opposed to one relating to reepithelialization<sup>[84]</sup>. Wounds from control db/db animals exhibited less aSMA staining when compared to those in the different scaffold treatment conditions <sup>[84]</sup>. This finding bears relevance in the context of translation to the human, as it focuses on aspects of wound chronicity in actually modeled in db/db mice. If a fundamental contractile defect exists in the db/db mouse, then any inferences made with regard to slower closure kinetics are confounded by the fact that epithelial cells have to migrate further to close the wound. Considering the findings of the present study, we were able to partially validate the db/db MSc. Thesis – April 2024 107

mouse as a model of hypoxic healing. However, we postulate this model to be defective in a sense that although it exhibits aspects of hypoxic healing, but not more than the wild type. Our preliminary evidence suggests that by day 12 the core of the wild type wound begins to exhibit hypoxia resolution (Appendix 4.8), however with an n=2 for this timepoint, further staining is needed to definitively confirm this. Our findings of increased neutrophil load at day 7 validate this model as one of delayed inflammatory resolution, however, we postulate this to be a manifestation of an overall delay in inflammatory cell recruitment and/or clearance and not of an overall greater recruitment of neutrophils in the diabetic wound. Similarly, this model exhibited an angiogenic deficit at day 3, validating it as a model of insufficient re-vascularization.

A secondary aim of this project was validation of whether neutrophil counts remain unchanged through different stages of healing as described in human data; comparably high levels of neutrophils in healed wounds up to 4-12 weeks old <sup>[82]</sup>. We report a failure of this model to recapitulate this aspect of human healing. We also performed preliminary experiments is porcine wounded skin tissue to see if it would exhibit similar levels of neutrophil infiltration at days 7 and 28. We observed neutrophils infiltration at the wound edge at day 7, however, their numbers qualitatively were diminished at day 28 (Figure 11). To compare this data to the human chronic wound, we carried out neutrophil elastase staining in chronic wounds of patients ranging from 47-87 years old. We observed a pronounced pattern of neutrophil infiltration at the edges of human chronic wounds, as well as immediately under the eschar (Figure 10).

It must be considered that pathologies can take decades to develop in the human, while in the mouse, researchers are trying to model these processes in a span of 0-1.5 years. It is entirely logical to presume the possibility of insufficient time for diabetes related pathologies to manifest in comparable levels to the human after such a short time. It must also be noted that after such a long period of time of mouse focused research, it is possible that the db/db model has undergone epigenetic changes, potentially confounding any data coming from work with this model.

#### **3.2.7 Future Directions**

Due to the fact that altered hypoxia adaptation could plausibly be implicated in every area of interest in our project, we believe that future work should further focus on transcription factors known to be upregulated in hypoxia and their target genes. Aberrant hypoxic signaling in the context of hyperglycemia, persistent inflammation and vascular deficiency are all well described cornerstones underpinning the commonly reported diabetes associated pathologies <sup>[115,121,122]</sup>. Although all these pathological factors represent promising therapeutic targets, if sufficient understanding is not gained in how cells temporally perceive and adapt to hypoxia in the human, any model tested treatments designed to potentially address them are unlikely to progress to the clinical setting.

An additional caveat is that the great majority of chronic wounds are associated with conditions that tend to manifest in older human demographics, including vascular disease, diabetes, and age-related changes in the inflammatory response <sup>[110]</sup>. According to Jackson Laboratory, mice ranging from 18-24 months correlate with humans ranging from 56-69 years of age <sup>[142]</sup>. The mice used in the present study were approximately 5 months old. Thus, *MSc. Thesis – April 2024* 109

an inherent limitation of our study is the age of our mice, as the data obtained from them would likely correlate more closely with a young human, thus potentially limiting the translatory value of our data to impaired wound healing in the younger adult. To closer mimic what is seen in the clinical setting with regard to the DFU, we suggest that future studies should consider using an aged mouse model of at least 1-1.5 years in order to more accurately recapitulate any wound healing processes occurring in the aged human, that could be masked if using a younger, healthier mouse.

Additionally, we propose that future work should focus on HIF-1 $\alpha$  and its activation in the earliest stages of wound healing, with the aim of elucidating what cell populations are chiefly responsible for early hypoxic adaptation. Hyperglycemia has also been extensively discussed as a factor inhibiting HIF-1 $\alpha$  stability, thus future studies should aim to expand on this finding and attempt to elucidate whether treatment in aged mice with insulin, has the potential to attenuate the negative effects of hyperglycemia resulting aberrant HIF-1 $\alpha$ signaling. If translation is the ultimate goal, an ideal study would utilize an aged wild-type and db/db mouse model of at least 1-1.5 years old, within the db/db cohort there should be a treatment group with insulin to attempt to elucidate if the negative impacts of hyperglycemia on hypoxic adaptation could potentially be reversed. Immunohistochemical co-labeling could then be employed to co-stain cell populations of interest for Hypoxyprobe<sup>TM</sup> and HIF-1 $\alpha$ activity, across timepoints corresponding to known wound healing events such as day 1, 37 and 21 to further elucidate the temporal activation profile of this transcription factor and attempt to correlate it with wound healing outcomes such as closure kinetics, inflammation and degree of revascularization. Western Blotting could be used to confirm the protein levels of HIF-1a, VEGF and any other proteins of interest. To complete this study, RT-qPCR could be used to confirm RNA levels of proteins of interest. This data could further be

MSc. Thesis – April 2024

supplemented with *in vitro* studies, such as growing murine aged cells like fibroblasts, keratinocytes, neutrophils and endothelial progenitors in varying hypoxic and glucose conditions to further elucidate their adaptative response to hypoxia under hyperglycemic conditions. An additional future direction for research could be attempting to elucidate the factors responsible for the macrophage polarization shift from M1 to M2. This concept is relatively new, however it is evident that there exist distinct macrophage phenotypes in the wound <sup>[136]</sup>. There are recognized shortcomings in techniques available to conclusively identify the two distinct phenotypes <sup>[136]</sup>. We believe it is worthwile to investigate additional markers that could aid researchers in positively identifying these phenotypes. Additionally, further work is needed to definitively characterize the factors responsible for the polarization shift, as such findings could bear significant implications for wound healing therapeutics.

Refocusing on the issue of clinical translation, we argue that chronic wound studies should ideally be carried out with close involvement of clinicians who have access to human chronic wound tissue. Due to the known differences in healing between mice and humans, any murine data no matter how promising, should be interpreted with caution and ideally validated and replicated in human tissue. Current work in our lab is performed with close involvement of clinicians in the field. Such co-operation allows us to gain access to and analyze tissues from the DFU prior to the terminal endpoint of amputation. Our evidence suggests that by the time of amputation, the wound profile is so dysregulated that any intervention at that stage likely futile. Clinical wound healing research is often limited by ethical considerations, affording most academic investigators only the terminal endpoint from which to draw conclusions. With the help of our clinician partners, we have been able to gain access to debridement tissues, sometimes from the same patient, allowing us to *MSc. Thesis – April 2024* 111

investigate and track any potential changes in cellular processes longitudinally. We propose supplementing the above-described murine study with age matched cell culture studies, as we have shown that it is possible to isolate and culture cells from diabetic debridement tissues. We propose these future studies to be undertaken with the ultimate goal of further elucidating the temporal profile of cellular adaptive responses to hypoxia, the effect of hyperglycemia in the context of aging, and any effects on inflammatory and angiogenic responses. Additional work should also be undertaken in the area of macrophage polarization and the vascular plasticity window, as these areas represent exciting, promising areas of study for the DFU. Although hypoxia represents an attractive therapeutic target, we believe it is premature to attempt to tackle it head on, without a prior detailed understanding of what cells are most responsive to it, and at what stages in wound healing it plays the most prominent role.

#### 3.2.8 References:

- [1] R. Nunan, K. G. Harding, P. Martin, DMM Disease Models and Mechanisms 2014, 7, 1205.
- [2] T. Mustoe, *The American Journal of Surgery* **2004**, *187*, S65.
- [3] M. A. Fonder, G. S. Lazarus, D. A. Cowan, B. Aronson-Cook, A. R. Kohli, A. J. Mamelak, *J Am Acad Dermatol* **2008**, *58*, 185.
- [4] T. Hurd, *Wounds International* **2013**, *4*, 13.
- [5] P. Driscoll, "Wound prevalence and wound management," can be found under http://blog.mediligence.com/2013/01/29/wound-prevalence-and-wound-management-2012-2020., 2012.
- [6] C. S. Garwood, J. S. Steinberg, P. J. Kim, *Clin Podiatr Med Surg* **2015**, *32*, 121.
- [7] D. Baltzis, I. Eleftheriadou, A. Veves, *Adv Ther* **2014**, *31*, 817.
- [8] A. Alavi, R. G. Sibbald, D. Mayer, L. Goodman, M. Botros, D. G. Armstrong, K. Woo, T. Boeni,
  E. A. Ayello, R. S. Kirsner, J Am Acad Dermatol 2014, 70, 1.e1.
- [9] W. J. Jeffcoate, K. G. Harding, *Lancet* **2003**, *361*, 1545.
- [10] D. Martins-Mendes, M. Monteiro-Soares, E. J. Boyko, M. Ribeiro, P. Barata, J. Lima, R. Soares, *J Diabetes Complications* **2014**, *28*, 632.

MSc. Thesis – April 2024

- [11] S. C. Wu, V. R. Driver, J. S. Wrobel, D. G. Armstrong, *Foot Ulcers in the Diabetic Patient, Prevention and Treatment*, **2007**.
- [12] V. Falanga, *The Lancet* **2005**, *366*, 1736.
- [13] L. Prompers, N. Schaper, J. Apelqvist, M. Edmonds, E. Jude, D. Mauricio, L. Uccioli, V.
  Urbancic, K. Bakker, P. Holstein, A. Jirkovska, A. Piaggesi, G. Ragnarson-Tennvall, H. Reike, M.
  Spraul, K. Van Acker, J. Van Baal, F. Van Merode, I. Ferreira, M. Huijberts, *Diabetologia* 2008, 51, 747.
- [14] M. A. Fonder, G. S. Lazarus, D. A. Cowan, B. Aronson-Cook, A. R. Kohli, A. J. Mamelak, *J Am Acad Dermatol* **2008**, *58*, 185.
- [15] T. Mustoe, *The American Journal of Surgery* **2004**, *187*, S65.
- [16] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [17] T. Mustoe, *The American Journal of Surgery* **2004**, *187*, S65.
- [18] R. Diegelmann, M. Evans, *Frontier Biosciences* **2004**, 283.
- [19] R. Goldman, Adv Skin Wound Care 2004, 17, 24.
- [20] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [21] J. Li, J. Chen, R. Kirsner, *Clin Dermatol* **2007**, *25*, 9.
- [22] R. F. Diegelmann, M. C. Evans, Front Biosci 2004, 9, 283.
- [23] S. Nagaraja, A. Wallqvist, J. Reifman, A. Y. Mitrophanov, J Immunol 2014, 192, 1824.
- [24] D. Baltzis, I. Eleftheriadou, A. Veves, *Adv Ther* **2014**, *31*, 817.
- [25] R. Blakytny, E. Jude, *Diabet Med* **2006**, *23*, 594.
- [26] M. Peppa, P. Stavroulakis, S. A. Raptis, *Wound Repair Regen* **2009**, *17*, 461.
- [27] R. Gary Sibbald, K. Y. Woo, *Diabetes Metab Res Rev* **2008**, *24 Suppl* 1, S25.
- [28] J. Berlanga-Acosta, G. S. Schultz, E. López-Mola, G. Guillen-Nieto, M. García-Siverio, L. Herrera-Martínez, *Biomed Res Int* 2013, 2013, 1.
- [29] R. F. Diegelmann, M. C. Evans, Front Biosci 2004, 9, 283.
- [30] R. Zhao, H. Liang, E. Clarke, C. Jackson, M. Xue, Int J Mol Sci 2016, 17, DOI 10.3390/ijms17122085.
- [31] L. H. Toledo-Pereyra, F. Lopez-Neblina, A. H. Toledo, Ann Transplant 2004, 9, 81.
- [32] H. K. Eltzschig, P. Carmeliet, N Engl J Med **2011**, 364, 656.
- [33] W. Ertel, M. H. Morrison, A. Ayala, I. H. Chaudry, *Am J Physiol* **1995**, *269*, R160.

- [34] G. Hartmann, M. Tschöp, R. Fischer, C. Bidlingmaier, R. Riepl, K. Tschöp, H. Hautmann, S. Endres, M. Toepfer, *Cytokine* 2000, 12, 246.
- [35] C. Brigati, B. Banelli, A. di Vinci, I. Casciano, G. Allemanni, A. Forlani, L. Borzì, M. Romani, Mediators Inflamm 2010, 2010, 263914.
- [36] H. Rosen, J. R. Crowley, J. W. Heinecke, *J Biol Chem* **2002**, *277*, 30463.
- [37] C. W. Baldridge, R. W. Gerard, *American Journal of Physiology-Legacy Content* **1932**, *103*, 235.
- [38] J. Karhausen, V. H. Haase, S. P. Colgan, *Cell Cycle* **2005**, *4*, 256.
- [39] A. F. McGettrick, L. A. J. O'Neill, *Cell Metab* **2020**, *32*, 524.
- [40] E. Kolaczkowska, P. Kubes, *Nat Rev Immunol* **2013**, *13*, 159.
- S. R. Walmsley, C. Print, N. Farahi, C. Peyssonnaux, R. S. Johnson, T. Cramer, A. Sobolewski, A. M. Condliffe, A. S. Cowburn, N. Johnson, E. R. Chilvers, *J Exp Med* 2005, *201*, 105.
- [42] T. N. Demidova-Rice, M. R. Hamblin, I. M. Herman, Adv Skin Wound Care 2012, 25, 304.
- [43] T. J. Koh, L. A. DiPietro, *Expert Rev Mol Med* **2011**, *13*, e23.
- [44] A. J. Meszaros, J. S. Reichner, J. E. Albina, *J Immunol* **2000**, *165*, 435.
- [45] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [46] A. J. Meszaros, J. S. Reichner, J. E. Albina, *J Leukoc Biol* **1999**, *65*, 35.
- [47] J. M. Daley, J. S. Reichner, E. J. Mahoney, L. Manfield, W. L. Henry, B. Mastrofrancesco, J. E. Albina, J Immunol 2005, 174, 2265.
- [48] J. S. Savill, A. H. Wyllie, J. E. Henson, M. J. Walport, P. M. Henson, C. Haslett, J Clin Invest 1989, 83, 865.
- [49] V. A. Fadok, D. L. Bratton, A. Konowal, P. W. Freed, J. Y. Westcott, P. M. Henson, J Clin Invest 1998, 101, 890.
- [50] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [51] J. B. Acosta, D. G. del Barco, D. C. Vera, W. Savigne, P. Lopez-Saura, G. Guillen Nieto, G. S. Schultz, Int Wound J 2008, 5, 530.
- [52] S. Khanna, S. Biswas, Y. Shang, E. Collard, A. Azad, C. Kauh, V. Bhasker, G. M. Gordillo, C. K. Sen, S. Roy, *PLoS One* 2010, *5*, e9539.
- [53] T. N. Demidova-Rice, J. T. Durham, I. M. Herman, Adv Wound Care (New Rochelle) 2012, 1, 17.
- [54] M. G. Tonnesen, X. Feng, R. A. Clark, J Investig Dermatol Symp Proc 2000, 5, 40.
- [55] D. Fukumura, R. K. Jain, *APMIS* **2008**, *116*, 695.
- MSc. Thesis April 2024

- [56] K. E. Johnson, T. A. Wilgus, Adv Wound Care (New Rochelle) 2014, 3, 647.
- [57] A. M. Szpaderska, J. D. Zuckerman, L. A. DiPietro, *J Dent Res* **2003**, *82*, 621.
- [58] K. K. Hirschi, P. A. D'Amore, *Cardiovasc Res* **1996**, *32*, 687.
- [59] G. Allt, J. G. Lawrenson, *Cells Tissues Organs* **2001**, *169*, 1.
- P. Lindblom, H. Gerhardt, S. Liebner, A. Abramsson, M. Enge, M. Hellstrom, G. Backstrom, S. Fredriksson, U. Landegren, H. C. Nystrom, G. Bergstrom, E. Dejana, A. Ostman, P. Lindahl, C. Betsholtz, *Genes Dev* 2003, *17*, 1835.
- [61] M. L. Broadhead, S. P. Becerra, P. F. M. Choong, C. R. Dass, Growth Factors 2010, 28, 280.
- [62] M. S. Wietecha, M. J. Król, E. R. Michalczyk, L. Chen, P. G. Gettins, L. A. Dipietro, *Am J Physiol Heart Circ Physiol* **2015**, *309*, H812.
- [63] M. S. Wietecha, M. J. Król, E. R. Michalczyk, L. Chen, P. G. Gettins, L. A. DiPietro, *Am J Physiol Heart Circ Physiol* **2015**, *309*, H812.
- [64] M. S. Wietecha, W. L. Cerny, L. A. DiPietro, *Curr Top Microbiol Immunol* **2013**, *367*, 3.
- [65] C. A. Staton, M. Valluru, L. Hoh, M. W. R. Reed, N. J. Brown, *Br J Dermatol* **2010**, *163*, 920.
- [66] Y. Brudno, A. B. Ennett-Shepard, R. R. Chen, M. Aizenberg, D. J. Mooney, *Biomaterials* 2013, 34, 9201.
- [67] P. Carmeliet, *Nature* **2005**, *438*, 932.
- [68] M. M. Hickey, M. C. Simon, *Curr Top Dev Biol* **2006**, *76*, 217.
- [69] G. C. Gurtner, S. Werner, Y. Barrandon, M. T. Longaker, *Nature* 2008, 453, 314.
- [70] L. A. DiPietro, *J Leukoc Biol* **2016**, *100*, 979.
- [71] A. M. Isidori, M. A. Venneri, D. Fiore, *J Endocrinol Invest* **2016**, *39*, 1235.
- [72] H. Kämpfer, J. Pfeilschifter, S. Frank, *Expressional Regulation of Angiopoietin-1 and-2 and the Tie-1 and-2 Receptor Tyrosine Kinases during Cutaneous Wound Healing: A Comparative Study of Normal and Impaired Repair*, **2001**.
- [73] C. Hellberg, A. Ostman, C.-H. Heldin, *Recent Results Cancer Res* 2010, 180, 103.
- [74] H. D. Beer, M. T. Longaker, S. Werner, *J Invest Dermatol* **1997**, *109*, 132.
- [75] R. D. Galiano, O. M. Tepper, C. R. Pelo, K. A. Bhatt, M. Callaghan, N. Bastidas, S. Bunting, H. G. Steinmetz, G. C. Gurtner, Am J Pathol 2004, 164, 1935.
- [76] W. Qi, C. Yang, Z. Dai, D. Che, J. Feng, Y. Mao, R. Cheng, Z. Wang, X. He, T. Zhou, X. Gu, L. Yan,
  X. Yang, J.-X. Ma, G. Gao, *Diabetes* 2015, *64*, 1407.
- [77] N. Urao, U. A. Okonkwo, M. M. Fang, Z. W. Zhuang, T. J. Koh, L. A. DiPietro, *Microvasc Res* 2016, 106, 57.

- [78] S. Sangiorgi, A. Manelli, M. Reguzzoni, M. Ronga, M. Protasoni, C. Dell'Orbo, Anat Rec (Hoboken) 2010, 293, 1639.
- [79] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [80] E. L. Campbell, W. J. Bruyninckx, C. J. Kelly, L. E. Glover, E. N. McNamee, B. E. Bowers, A. J. Bayless, M. Scully, B. J. Saeedi, L. Golden-Mason, S. F. Ehrentraut, V. F. Curtis, A. Burgess, J. F. Garvey, A. Sorensen, R. Nemenoff, P. Jedlicka, C. T. Taylor, D. J. Kominsky, S. P. Colgan, *Immunity* **2014**, *40*, 66.
- [81] S. R. Walmsley, C. Print, N. Farahi, C. Peyssonnaux, R. S. Johnson, T. Cramer, A. Sobolewski, A. M. Condliffe, A. S. Cowburn, N. Johnson, E. R. Chilvers, *Journal of Experimental Medicine* 2005, *201*, 105.
- [82] R. J. Butterworth, M. D. Thesis, THE HISTOLOGY OF HUMAN GRANULATING WOUNDS, 1992.
- [83] S. Schreml, R. M. Szeimies, L. Prantl, S. Karrer, M. Landthaler, P. Babilas, British Journal of Dermatology 2010, 163, 257.
- [84] C. G. Elliott, J. Wang, J. T. Walker, S. Michelsons, J. Dunmore-Buyze, M. Drangova, A. Leask, D.
  W. Hamilton, *Tissue Eng Part A* **2019**, *25*, 1326.
- [85] D. W. Siemann, *Cancer Treat Rev* **2011**, *37*, 63.
- [86] D. J. Kominsky, E. L. Campbell, S. P. Colgan, *The Journal of Immunology* **2010**, *184*, 4062.
- [87] F. L. Cousins, A. A. Murray, J. P. Scanlon, P. T. K. Saunders, *BMC Res Notes* 2016, 9, DOI 10.1186/s13104-016-1842-8.
- [88] S. Kang, D. Lee, B. E. Theusch, C. J. Arpey, T. J. Brennan, *Wound Repair and Regeneration* **2013**, *21*, 730.
- [89] B. Bedogni, M. B. Powell, *Pigment Cell Melanoma Res* 2009, 22, 166.
- [90] V. Flacher, F. Sparber, C. H. Tripp, N. Romani, P. Stoitzner, in *Cancer Immunology, Immunotherapy*, **2009**, pp. 1137–1147.
- [91] A. Stachura, I. Khanna, P. Krysiak, W. Paskal, P. Włodarski, Int J Mol Sci 2022, 23, DOI 10.3390/ijms23158621.
- [92] A. Bavelloni, G. Ramazzotti, A. Poli, M. Piazzi, E. Focaccia, W. Blalock, I. Faenza, *Anticancer Res* **2017**, *37*, 6511.
- S. Narayanan, S. Eliasson Angelstig, C. Xu, J. Grünler, A. Zhao, W. Zhu, N. Xu Landén, M.
  Ståhle, J. Zhang, M. Ivan, R. G. Maltesen, I. R. Botusan, N. Rajamand Ekberg, X. Zheng, S. B.
  Catrina, *Commun Biol* 2020, *3*, DOI 10.1038/s42003-020-01495-y.
- [94] S.-B. Catrina, K. Okamoto, T. Pereira, K. Brismar, L. Poellinger, *Hyperglycemia Regulates Hypoxia-Inducible Factor-1 Protein Stability and Function*, **2004**.
- [95] U. A. Okonkwo, L. Chen, D. Ma, V. A. Haywood, M. Barakat, N. Urao, L. A. DiPietro, *PLoS One* 2020, 15, DOI 10.1371/journal.pone.0231962.

MSc. Thesis – April 2024

- [96] J. Todoric, M. Löffler, J. Huber, M. Bilban, M. Reimers, A. Kadl, M. Zeyda, W. Waldhäusl, T. M. Stulnig, *Diabetologia* 2006, 49, 2109.
- [97] S. W. Hartwell, SURGICAL WOUNDS IN HUMAN BEINGS A HISTOLOGIC STUDY OF HEALING WITH PRACTICAL APPLICATIONS : II. FIBROUS HEALING, n.d.
- [98] A. F. McGettrick, L. A. J. O'Neill, *Cell Metab* **2020**, *32*, 524.
- [99] A. C. D. O. Gonzalez, Z. D. A. Andrade, T. F. Costa, A. R. A. P. Medrado, An Bras Dermatol 2016, 91, 614.
- [100] T. Cramer, Y. Yamanishi, B. E. Clausen, I. Förster, R. Pawlinski, N. Mackman, V. H. Haase, R. Jaenisch, M. Corr, V. Nizet, G. S. Firestein, H.-P. Gerber, N. Ferrara, R. S. Johnson, *HIF-1α Is Essential for Myeloid Cell-Mediated Inflammation*, **n.d.**
- [101] A. P. Sawaya, R. C. Stone, S. R. Brooks, I. Pastar, I. Jozic, K. Hasneen, K. O'Neill, S. Mehdizadeh, C. R. Head, N. Strbo, M. I. Morasso, M. Tomic-Canic, *Nat Commun* **2020**, *11*, DOI 10.1038/s41467-020-18276-0.
- [102] L. A. DiPietro, J Leukoc Biol **2016**, 100, 979.
- [103] J. Wang, D. Tinney, M. Grynyshyn, J. G. Pickering, A. Power, L. Dubois, D. W. Hamilton, Wound Repair and Regeneration 2023, 31, 321.
- [104] **N.d.**
- [105] W. X. Hong, M. S. Hu, M. Esquivel, G. Y. Liang, R. C. Rennert, A. McArdle, K. J. Paik, D. Duscher, G. C. Gurtner, H. P. Lorenz, M. T. Longaker, *Adv Wound Care (New Rochelle)* 2014, *3*, 390.
- [106] H. Thangarajah, D. Yao, E. I. Chang, Y. Shi, L. Jazayeri, I. N. Vial, R. D. Galiano, X.-L. Du, R. Grogan, M. G. Galvez, M. Januszyk, M. Brownlee, G. C. Gurtner, *The Molecular Basis for Impaired Hypoxia-Induced VEGF Expression in Diabetic Tissues*, n.d.
- [107] L. Gould, P. Abadir, H. Brem, M. Carter, T. Conner-Kerr, J. Davidson, L. Dipietro, V. Falanga, C. Fife, S. Gardner, E. Grice, J. Harmon, W. R. Hazzard, K. P. High, P. Houghton, N. Jacobson, R. S. Kirsner, E. J. Kovacs, D. Margolis, F. McFarland Horne, M. J. Reed, D. H. Sullivan, S. Thom, M. Tomic-Canic, J. Walston, J. A. Whitney, J. Williams, S. Zieman, K. Schmader, J Am Geriatr Soc 2015, 63, 427.
- [108] L. Pradhan, C. Nabzdyk, N. D. Andersen, F. W. LoGerfo, A. Veves, *Expert Rev Mol Med* 2009, 11, DOI 10.1017/S1462399409000945.
- [109] E. Groppa, A. Colliva, R. Vuerich, T. Kocijan, S. Zacchigna, Int J Mol Sci 2020, 21, 1.
- [110] L. Gould, P. Abadir, H. Brem, M. Carter, T. Conner-Kerr, J. Davidson, L. Dipietro, V. Falanga, C. Fife, S. Gardner, E. Grice, J. Harmon, W. R. Hazzard, K. P. High, P. Houghton, N. Jacobson, R. S. Kirsner, E. J. Kovacs, D. Margolis, F. McFarland Horne, M. J. Reed, D. H. Sullivan, S. Thom, M. Tomic-Canic, J. Walston, J. A. Whitney, J. Williams, S. Zieman, K. Schmader, *J Am Geriatr Soc* 2015, *63*, 427.
- [111] J. Wang, Cell Tissue Res **2018**, 371, 531.
- MSc. Thesis April 2024

- [112] A. J. Varghese, S. Gulyas, J. K. Mohindra, *Cancer Res* **1976**, *36*, 3761.
- [113] S. Kang, D. Lee, B. E. Theusch, C. J. Arpey, T. J. Brennan, Wound Repair and Regeneration 2013, 21, 730.
- [114] F. L. Cousins, A. A. Murray, J. P. Scanlon, P. T. K. Saunders, BMC Res Notes 2016, 9, DOI 10.1186/s13104-016-1842-8.
- [115] S.-B. Catrina, K. Okamoto, T. Pereira, K. Brismar, L. Poellinger, *Hyperglycemia Regulates Hypoxia-Inducible Factor-1 Protein Stability and Function*, **2004**.
- [116] Y. Guan, H. Niu, Z. Liu, Y. Dang, J. Shen, M. Zayed, L. Ma, J. Guan, H E A L T H A N D M E D I C I N E Sustained Oxygenation Accelerates Diabetic Wound Healing by Promoting Epithelialization and Angiogenesis and Decreasing Inflammation, 2021.
- [117] R. M. Leach, D. F. Treacher, ABC of Oxygen Oxygen Transport-2. Tissue Hypoxia, n.d.
- [118] H. Brem, O. Stojadinovic, R. F. Diegelmann, H. Entero, B. Lee, I. Pastar, M. Golinko, H. Rosenberg, M. Tomic-Canic, *Molecular Medicine* 2007, 13, 30.
- [119] Y. Guan, H. Niu, Z. Liu, Y. Dang, J. Shen, M. Zayed, L. Ma, J. Guan, H E A L T H A N D M E D I C I N E Sustained Oxygenation Accelerates Diabetic Wound Healing by Promoting Epithelialization and Angiogenesis and Decreasing Inflammation, 2021.
- [120] S. Narayanan, S. Eliasson Angelstig, C. Xu, J. Grünler, A. Zhao, W. Zhu, N. Xu Landén, M. Ståhle, J. Zhang, M. Ivan, R. G. Maltesen, I. R. Botusan, N. Rajamand Ekberg, X. Zheng, S. B. Catrina, *Commun Biol* 2020, *3*, DOI 10.1038/s42003-020-01495-y.
- [121] H. Thangarajah, D. Yao, E. I. Chang, Y. Shi, L. Jazayeri, I. N. Vial, R. D. Galiano, X.-L. Du, R. Grogan, M. G. Galvez, M. Januszyk, M. Brownlee, G. C. Gurtner, *The Molecular Basis for Impaired Hypoxia-Induced VEGF Expression in Diabetic Tissues*, n.d.
- [122] T. Cramer, Y. Yamanishi, B. E. Clausen, I. Förster, R. Pawlinski, N. Mackman, V. H. Haase, R. Jaenisch, M. Corr, V. Nizet, G. S. Firestein, H.-P. Gerber, N. Ferrara, R. S. Johnson, *HIF-1α Is Essential for Myeloid Cell-Mediated Inflammation*, **n.d.**
- [123] J. Song, X. Yang, L.-J. Yan, *Hypoxia* **2019**, *Volume* 7, 33.
- [124] S. Taneja, S. W. Vetter, E. Leclerc, Int J Mol Sci 2021, 22, DOI 10.3390/ijms22158153.
- [125] V. P. Singh, A. Bali, N. Singh, A. S. Jaggi, *Korean Journal of Physiology and Pharmacology* **2014**, *18*, 1.
- [126] G. Zaccagnini, S. Greco, C. Voellenkle, C. Gaetano, F. Martelli, Antioxid Redox Signal 2022, 36, 685.
- [127] A. Bavelloni, G. Ramazzotti, A. Poli, M. Piazzi, E. Focaccia, W. Blalock, I. Faenza, Anticancer Res 2017, 37, 6511.
- [128] S. Biswas, S. Roy, J. Banerjee, S. R. A. Hussain, S. Khanna, G. Meenakshisundaram, P. Kuppusamy, A. Friedman, C. K. Sen, *Proc Natl Acad Sci U S A* **2010**, *107*, 6976.

- [129] A. P. Sawaya, R. C. Stone, S. R. Brooks, I. Pastar, I. Jozic, K. Hasneen, K. O'Neill, S. Mehdizadeh, C. R. Head, N. Strbo, M. I. Morasso, M. Tomic-Canic, *Nat Commun* **2020**, *11*, DOI 10.1038/s41467-020-18276-0.
- [130] C. Wetzler, H. K. Empfer, B. Stallmeyer, J. Pfeilschifter, S. Frank, Large and Sustained Induction of Chemokines during Impaired Wound Healing in the Genetically Diabetic Mouse: Prolonged Persistence of Neutrophils and Macrophages during the Late Phase of Repair, n.d.
- [131] T. A. Wilgus, S. Roy, J. C. McDaniel, Adv Wound Care (New Rochelle) 2013, 2, 379.
- [132] P. K. Barman, T. J. Koh, Front Cell Dev Biol 2020, 8, DOI 10.3389/fcell.2020.00528.
- [133] F. O. Quenum Zangbede, A. Chauhan, J. Sharma, B. B. Mishra, *Journal of Neuroscience* 2018, 38, 6737.
- [134] U. A. Okonkwo, L. Chen, D. Ma, V. A. Haywood, M. Barakat, N. Urao, L. A. DiPietro, *PLoS One* 2020, 15, DOI 10.1371/journal.pone.0231962.
- [135] H. Kämpfer, J. Pfeilschifter, S. Frank, Expressional Regulation of Angiopoietin-1 and-2 and the Tie-1 and-2 Receptor Tyrosine Kinases during Cutaneous Wound Healing: A Comparative Study of Normal and Impaired Repair, 2001.
- [136] N. Jetten, S. Verbruggen, M. J. Gijbels, M. J. Post, M. P. J. De Winther, M. M. P. C. Donners, Angiogenesis 2014, 17, 109.
- [137] S. Wang, G. Liu, Y. Li, Y. Pan, Front Immunol 2022, 13, DOI 10.3389/fimmu.2022.840029.
- [138] S. Elliot, T. C. Wikramanayake, I. Jozic, M. Tomic-Canic, *Journal of Investigative Dermatology* **2018**, *138*, 736.
- [139] P. Huynh, J. Phie, S. M. Krishna, J. Golledge, *BMJ Open Diabetes Res Care* 2020, 8, DOI 10.1136/bmjdrc-2019-000982.
- [140] R. C. Fang, Z. B. Kryger, D. W. Buck, M. De La Garza, R. D. Galiano, T. A. Mustoe, *Wound Repair and Regeneration* **2010**, *18*, 605.
- [141] L. Dunn, H. C. G. Prosser, J. T. M. Tan, L. Z. Vanags, M. K. C. Ng, C. A. Bursill, Journal of Visualized Experiments 2013, 2013, 1.
- [142] S. J. Jackson, N. Andrews, D. Ball, I. Bellantuono, J. Gray, L. Hachoumi, A. Holmes, J. Latcham, A. Petrie, P. Potter, A. Rice, A. Ritchie, M. Stewart, C. Strepka, M. Yeoman, K. Chapman, *Lab Anim* **2017**, *51*, 160.

## 4.0 Appendix

4.1 – Hypoxyprobe<sup>TM</sup> staining validation figure showing suboptimal detection with the default QuPATH vector, and improved detection with a manually set vector.

Hypoxyprobe Staining Validation

#### 1a)

Good Contrast - Pre Analysis



Default Vector DAB: 0.269 0.568 0.778



Good Contrast - Post Analysis



#### 1b)

Poor Contrast - Pre Analysis





#### Poor Contrast - Post Analysis





Poor Contrast - Pre Analysis







4.2 – Neutrophil elastase staining validation figure showing sub-optimal detection with default QuPATH vector, and improved detection with a manually set vector.

#### Neutrophil Elastase Staining Validation



1b)



Arrows denoting cells which would be excluded based on morphology

4.3 – CD31<sup>+</sup> staining validation figure showing sub-optimal detection with default vector, and improved detection with a manually set vector.

#### CD31 Staining Validation



Good Contrast - Pre Analysis



Default Vector DAB: 0.269 0.568 0.778

#### Good Contrast - Post Analysis



#### 1b)

Poor Contrast - Pre Analysis







1c)

#### Poor Contrast - Pre Analysis





Poor Contrast - Post Analysis



4.4 – Total cell counts at day 3 & 7 for wild type mice. Note the cell increase between days 3 and 7 and the slight trend toward more cell infiltration compared to db/db, especially in wound bed.

## Total Cell Count WT Hypoxyprobe - Day 3 and 7

WT - Left Edge Day 3/7 WT - Wound Bed Day 3/7

WT - Right Edge Day 3/7





4.5 – Total cell counts for days 3 & 7 for db/db mice. Note the cell increase between days 3 & 7.

### Total Cell Count Hypoxyprobe DB - Day 3 and 7



DB - Left Edge Day 3/7

- DB Wound Bed Day 3/7
- DB Right Edge Day 3/7

Appendix 4.6 – Figure showing "volcano like" infiltration pattern of hypoxic cells evident in both db/dbs and wild type mice

# Appendix 4.6 – Showing volcano like infiltration pattern of hypoxic cells from deep within tissue


Appendix 4.7 – Figure showing positive control for  $Hypoxyprobe^{TM}$  in the form of murine kidney.



Appendix 4.7 - Murine kidney Hypoxyprobe Positive Control

Appendix 4.8 – Figure showing diminished Hypoxyprobe<sup>TM</sup> staining at day 12 in the wild type. Note the wound core is still strongly staining for hypoxia in the db/db



## Appendix 4.8 - Evidence of Hypoxia Resolution at Day 12 in the Wild Type

Appendix 4.9 – Figure showing profound Hypoxyprobe<sup>TM</sup> staining in a tumor, which is known to have extremely dysregulated vasculature.

Appendix 4.9



Appendix 4.10 – Unwounded Murine Hypoxyprobe<sup>TM</sup> staining



Appendix 4.10 - Unwounded Murine Hypoxyprobe Stain

Appendix 4.11 – Animal Protocol Approval Letter

## Western 😽

AUP Number: 2020-142

PI Name: Hamilton, Doug

**AUP Title:** Influence of biomaterials and material physiochemical properties on cell behaviour in vitro and in vivo.

Official Notification of ACC Approval: A modification to Animal Use Protocol 2020-142 has been approved.

Please at this time review your AUP with your research team to ensure full understanding by everyone listed within this AUP.

As per your declaration within this approved AUP, you are obligated to ensure that:

1. This Animal Use Protocol is in compliance with:

- Western's Senate MAPP 7.12 [PDF]; and
- Applicable Animal Care Committee policies and procedures.
- 2. Prior to initiating any study-related activities—<u>as per institutional OH&S policies</u>—all individuals listed within this AUP who will be using or potentially exposed to hazardous materials will have:
  - Completed the appropriate institutional OH&S training;
  - Completed the appropriate facility-level training; and
  - Reviewed related (M)SDS Sheets.

## 5.0 Curriculum Vitae

Name:	Michael Roman Grynyshyn
Post-secondary	Western University, London
Education and	Ontario, Canada
Degrees:	2014-2020 B.A. Health Sciences
Honours and	
Awards:	Western Admission Scholarship, 2014
	Ontario Graduate Scholarship, 2020
Related Work	Research Assistant
Experience	Western University
	2019-2020

## **Publications**

- Hamilton DW, Walker JT, Tinney D, Grynyshyn M, El-Warrak A, Truscott E, Flynn LE. The pig as a model system for investigating the recruitment and contribution of myofibroblasts in skin healing. Wound Repair Regen. 2022 Jan;30(1):45-63. doi: 10.1111/wrr.12981. Epub 2021 Nov 6. PMID: 34708478.
  - Published January 2022, Wound Repair and Regeneration, Volume 30, Issue 1, pages 45-63
- Wang J, Tinney D, Grynyshyn M, Pickering JG, Power A, Dubois L, Hamilton DW. Microcirculation surrounding end-stage human chronic skin wounds is associated with endoglin/CD146/ALK-1 expression, endothelial cell proliferation and an absence of p16<sup>Ink4a</sup>. Wound Repair Regen. 2023 May-Jun;31(3):321-337. doi: 10.1111/wrr.13081. Epub 2023 Apr 13. PMID: 37017097.
  - Published April 2023, Wound Repair and Regeneration, Volume 31, Issue
    3. Pages 321-337